

Citation: Brain KL, Allison BJ, Niu Y, Cross CM, Itani N, Kane AD, et al. (2019) Intervention against hypertension in the next generation programmed by developmental hypoxia. PLoS Biol 17(1): e2006552. https://doi.org/10.1371/journal. pbio.2006552

Academic Editor: Cecilia Lo, University of Pittsburgh, United States of America

Received: May 3, 2018

Accepted: November 29, 2018

Published: January 22, 2019

Copyright: © 2019 Brain et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data that support the findings of this study are provided in Excel spreadsheet files (<u>S1 Data</u>).

Funding: British Heart Foundation <u>https://www.bhf.</u> org.uk/ (grant number RG/17/8/32924). The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. BBSRC. To KB Brain. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

Intervention against hypertension in the next generation programmed by developmental hypoxia

Kirsty L. Brain¹, Beth J. Allison¹, Youguo Niu^{1,2}, Christine M. Cross¹, Nozomi Itani¹, Andrew D. Kane¹, Emilio A. Herrera¹, Katie L. Skeffington¹, Kimberley J. Botting^{1,2}, Dino A. Giussani^{1,2}*

1 Department of Physiology, Development, and Neuroscience, University of Cambridge, Cambridge, United Kingdom, 2 Cambridge Cardiovascular Strategic Research Initiative, Cambridge, United Kingdom

* dag26@cam.ac.uk

Abstract

Evidence derived from human clinical studies and experimental animal models shows a causal relationship between adverse pregnancy and increased cardiovascular disease in the adult offspring. However, translational studies isolating mechanisms to design intervention are lacking. Sheep and humans share similar precocial developmental milestones in cardiovascular anatomy and physiology. We tested the hypothesis in sheep that maternal treatment with antioxidants protects against fetal growth restriction and programmed hypertension in adulthood in gestation complicated by chronic fetal hypoxia, the most common adverse consequence in human pregnancy. Using bespoke isobaric chambers, chronically catheterized sheep carrying singletons underwent normoxia or hypoxia (10% oxygen $[O_2]$) ± vitamin C treatment (maternal 200 mg.kg⁻¹ IV daily) for the last third of gestation. In one cohort, the maternal arterial blood gas status, the value at which 50% of the maternal hemoglobin is saturated with oxygen (P₅₀), nitric oxide (NO) bioavailability, oxidative stress, and antioxidant capacity were determined. In another, naturally delivered offspring were raised under normoxia until early adulthood (9 months). Lambs were chronically instrumented and cardiovascular function tested in vivo. Following euthanasia, femoral arterial segments were isolated and endothelial function determined by wire myography. Hypoxic pregnancy induced fetal growth restriction and fetal oxidative stress. At adulthood, it programmed hypertension by enhancing vasoconstrictor reactivity and impairing NO-independent endothelial function. Maternal vitamin C in hypoxic pregnancy improved transplacental oxygenation and enhanced fetal antioxidant capacity while increasing NO bioavailability, offsetting constrictor hyper-reactivity and replenishing endothelial function in the adult offspring. These discoveries provide novel insight into mechanisms and interventions against fetal growth restriction and adult-onset programmed hypertension in an animal model of complicated pregnancy in a species of similar temporal developmental milestones to humans.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: AngII, angiotensin II; ANOVA, analysis of variance: AAC. area above the curve: AWERB, Animal Welfare and Ethical Review Board; BMI, body mass index; BPD, biparietal diameter; CRL, crown rump length; dGA, days of gestation; EDHF, endothelium-derived hyperpolarizing factor; [Hb], hemoglobin concentration; HCO₃, bicarbonate; Hcy, homocysteine; INTAPP, international trial of antioxidants in the prevention of preeclampsia; LLL, lower limb length; L-NAME, $N(\omega)$ -nitro-L-arginine methyl ester; MetCh, methacholine; NO, nitric oxide; NOx, plasma concentrations of total $NO_3^- + NO_2^-$; O_2 , oxygen; P₅₀, PaO₂ value at which 50% of hemoglobin is saturated with oxygen; PaCO₂, arterial blood partial pressure of carbon dioxide; PaO₂, arterial blood partial pressure of oxygen; PBS, physiologic buffer solution; PE, phenylephrine; PG, prostaglandin; PM, post mortem; Sat Hb, arterial blood percentage saturation of hemoglobin with oxygen; ROS, reactive oxygen species; SEM, standard error of the mean; SNP, sodium nitroprusside; SOD, superoxide dismutase; percent Rmax, maximal vessel relaxation; VIP, vitamins in preeclampsia.

Author summary

Adverse conditions during pregnancy can increase the cardiovascular risk of the adult offspring. However, the mechanisms underlying these effects remain unclear, precluding the identification of candidate therapy. In this interventional study in sheep, a species of similar temporal developmental milestones to humans, we adopt an integrative approach, combining studies in vivo with those at the isolated organ, cellular, and molecular levels to investigate consequences of suboptimal pregnancy on the offspring at two stages of life: in the near-term fetus and in the adult. We show that developmental hypoxia, the most common outcome in human suboptimal pregnancy, slows fetal growth and programs high blood pressure in the sheep adult offspring. Maternal treatment with vitamin C in hypoxic pregnancy restored fetal growth and protected against adult-onset hypertension by improving transplacental oxygen delivery, enhancing fetal antioxidant capacity, and increasing the bioavailability of nitric oxide (NO) in the sheep adult offspring. Our discoveries highlight that when considering strategies to reduce the overall burden of heart disease, a much greater attention to prevention rather than treatment is required. Treatment should start as early as possible during the developmental trajectory, rather than waiting until adulthood when the disease process has become irreversible.

Introduction

Cardiovascular disease kills 1 in 3 people [1]. The annual costs for patient care and lost workforce due to heart disease are over US\$130 billion in the United States and Canada [2] and over £30 billion in the United Kingdom [3]. Therefore, cardiovascular disease is a significant problem imposing a substantial burden on every nation's health and wealth [4]. It is widely accepted that our genes interact with traditional lifestyle factors, such as smoking, obesity, and/or a sedentary lifestyle, to promote an increased risk of cardiovascular disease [5]. It is also established that the gene-environment interaction early in life may be just as, if not more, important in "programming" heart health and heart disease [6-8]. Evidence from human sibling-pair studies suggests that these relationships are causal, that they occur independently of genotype, and that they are significantly influenced by the quality of the intrauterine environment during pregnancy [9-12]. For instance, studies in Pima Indians showed a greater prevalence of type 2 diabetes in siblings born from pregnancies during which the mother had gestational diabetes compared to those whose mother did not [9]. Bariatric surgery to decrease the weight of obese women reduced the risk of obesity, insulin resistance, and raised blood pressure in children born after surgery compared to those born before surgery [10-12]. Therefore, these studies highlight a disproportionate risk of disease in offspring born from the same mother but under different in utero conditions, providing strong evidence in humans that the environment experienced during this critical period of development directly influences longterm cardiovascular health.

One of the most common outcomes of complicated pregnancy in humans is chronic fetal hypoxia leading to reduced fetal growth, as can occur during placental insufficiency, preeclampsia, or inflammatory conditions during pregnancy, such as in chorioamnionitis, gestational diabetes, or maternal obesity [13,14]. In humans, low birth weight is related to poor neonatal outcome [15] and endothelial dysfunction [16] and high blood pressure at adulthood [17]. In turn, increased blood pressure is associated with an increased risk of cardiovascular disease [18], with this risk being greatest in those who were smallest at birth but with the most accelerated weight gain in childhood [19]. To date, there is no cure for pregnancy complicated by chronic fetal hypoxia to protect against fetal growth restriction or programmed cardiovascular dysfunction in the offspring. Treatment options are restricted to monitoring surrogate measures of fetal hypoxia and fetal growth, ultimately ending up in elective delivery of the offspring [20]. This highlights the need for experimental studies addressing underlying mechanisms to identify plausible intervention.

Studies in animal models of adverse pregnancy have confirmed causality, reporting that oxidative stress during complicated pregnancy, including one involving chronic fetal hypoxia, may be a potential underlying mechanism [8,21-31]. Chronic fetal hypoxia is a powerful stimulus for reactive oxygen species (ROS) generation [8]. Under physiologic conditions, ROS are important mediators of a wide variety of cell functions, for instance, via signaling or by interacting with nitric oxide (NO) to provide a vascular oxidant tone [32-34]. However, excessive ROS and/or a fall in antioxidant defenses can lead to cellular oxidative stress and a fall in the bioavailability of NO, predisposing to cardiovascular dysfunction [8,31]. A small cluster of investigations including our own, mostly through studies of isolated hearts and vessels or echocardiography in rodents, has provided evidence for possible intervention with maternal treatment with antioxidants to protect against the ill effects of hypoxic pregnancy on the offspring [28–31,35,36]. However, when working with animal models of cardiovascular dysfunction before birth, the temporal profile of cardiovascular development between species is a highly important consideration for successful interventional translation to the human clinical situation. Rodents are altricial species, in which cardiovascular maturation continues past birth, becoming completed by the second week of postnatal life [28]. In contrast, sheep and humans share similar prenatal tempos of cardiovascular development [28] and some breeds of sheep, like Welsh Mountain, give birth primarily to singleton lambs of similar weight to term human babies. To date, no study has addressed maternal antioxidant intervention to protect against systemic cardiovascular dysfunction in the adult offspring in a human translational model of hypoxic pregnancy, such as in sheep, which additionally permits detailed cardiovascular analysis of in vivo mechanisms of action.

In the present study, we tested the antioxidant vitamin C, as it is widely supplemented in human populations. In an elegant study, Jackson and colleagues [37] reported that the capacity of the antioxidant vitamin C to scavenge O_2^- ex vivo and its ability to prevent O_2^- -induced impairment of endothelial function in vivo occurred at very different concentrations, requiring a much higher effective concentration in vivo. Therefore, the dose regimen used in the preset study was derived from previous studies in our laboratory, which achieved elevations in circulating ascorbate within the required range for vitamin C to act effectively in vivo in ovine pregnancy [33,34]. To put into context, the dose of vitamin C used in the present study was 8 times higher than the dose employed in human clinical trials to prevent preeclampsia [38]. Here, we tested the hypothesis using Welsh Mountain sheep that maternal treatment with vitamin C protects against fetal growth restriction and programmed hypertension in adulthood in gestation complicated by chronic fetal hypoxia. While experimental models, which affect uterine blood flow or placental function, impair both fetal nutrition and fetal oxygenation, the isolated effect of chronic hypoxia on the fetus can be best studied by exposing the ovine pregnancy to an environment of reduced oxygenation. We therefore created 4 isobaric chambers [39,40] able to maintain pregnant sheep for long periods of gestation (Fig 1A). Adopting an integrative approach at the in vivo, isolated organ, and molecular levels, we show that maternal treatment with vitamin C in ovine hypoxic pregnancy protects against both fetal growth restriction and hypertension in the adult offspring. Mechanisms underlying this antioxidant protection include improved transplacental oxygenation, enhanced endogenous antioxidant capacity, increased in vivo NO bioavailability, offset in vivo vasoconstrictor hyper-reactivity, and replenished endothelial function in the peripheral vasculature of the offspring (see



Fig 1. Maternal data. 1A. Isobaric hypoxic chamber; maternal variables measured during the experimental period are the following: 1B, maternal atmospheric oxygen exposure; 1C, arterial blood partial pressure of oxygen (PaO₂); 1D, Sat Hb; 1E, plasma NOx during the experimental period and at PM; 1F, maternal daily food consumption; 1G, maternal weight gain during the experimental period; 1H, plasma vitamin C during the experimental period and at PM, and 1I, the P₅₀ value of the relationship between the maternal PaO₂ plotted against the Sat Hb. Values are mean ± SEM for all data. Groups are N (open symbols, *n* = 7–21), H (black/grey symbols, *n* = 7–22), HC (red symbols, *n* = 6–19), and NC (blue symbols, *n* = 9–18). Significant (*P* < 0.05) differences are the following: *versus normoxia, † versus baseline, ‡H versus HC, and two-way repeated-measures ANOVA with posthoc Tukey test (1B–1F and 1H) or Student *t* test for unpaired data (1I). H, hypoxia; HC, hypoxia with vitamin C; N, normoxia; NC, normoxia with vitamin C; NOx, plasma concentrations of total NO₃⁻ + NO₂⁻; P₅₀, PaO₂ value at which 50% of hemoglobin is saturated with oxygen; PaO₂, arterial blood partial pressure of oxygen; PM, post mortem; Sat Hb, arterial blood percentage saturation of hemoglobin with oxygen.

https://doi.org/10.1371/journal.pbio.2006552.g001



Fig 2. Summary illustration. Hypoxic pregnancy in sheep increases fetal oxidative stress, reduces fetal growth, and programs NO-independent endothelial dysfunction, vasoconstrictor hyper-reactivity, and hypertension in the adult offspring. Maternal treatment with vitamin C in hypoxic pregnancy increases transplacental oxygenation and fetal antioxidant capacity. This protects fetal growth and restores NO bioavailability, normalizing peripheral vascular function and arterial blood pressure in the adult offspring.

https://doi.org/10.1371/journal.pbio.2006552.g002

Summary illustration, Fig 2). By studying a species of similar developmental milestones to humans, the study therefore presents a conceptual advance to this field of research. It allows not only in vivo investigation of basal and stimulated cardiovascular physiology in the chronically instrumented adult offspring in an animal model that closely recapitulates human pregnancies involving fetal oxygen insufficiency, but it also provides a spring board toward human clinical translation, offering viable treatment options for the developmental origins of hypertension.

Results

Prenatal hypoxia ± antioxidant treatment: Effects on the mother, placenta, and fetus

In control ewes undergoing normoxic pregnancy, maternal arterial blood gases and pH remained unaltered from baseline until 135 days of gestation (dGA) (Fig 1B-1D and Table 1). However, in these ewes, there was a significant increase in maternal bicarbonate (HCO_3^-) on

PLOS BIOLOGY

Table 1. Maternal blood gas, acid-base, and metabolic status. Values are mean \pm SEM for the maternal arterial pH, PaCO₂, PaO₂, Sat Hb, HCO₃⁻, and [Hb], measured at baseline (102 until 105 dGA) and at set intervals during the experimental period (106 until 138 dGA). Groups are normoxia, hypoxia, hypoxia with vitamin C, and normoxia with vitamin C. N = 19 for all groups. Significant (P < 0.05) differences are *versus normoxia, †H versus HC, * versus baseline, and two-way repeated-measures ANOVA with posthoc Tukey test.

		Baseline		Hypoxia/Normoxia ± Vitamin C						
		102 dGA	105 dGA	106 dGA	110 dGA	115 dGA	120 dGA	125 dGA	130 dGA	135 dGA
рН	Normoxia	7.50 ± 0.02	7.50 ± 0.01	7.51 ± 0.01	7.50 ± 0.01	7.49 ± 0.01	7.49 ± 0.01	7.49 ± 0.01	7.48 ± 0.01	7.50 ± 0.01
	Hypoxia	7.50 ± 0.01	7.50 ± 0.01	$7.54 \pm 0.01^{*}$ ‡	7.51 ± 0.01	7.51 ± 0.01	7.51 ± 0.01	7.50 ± 0.01	7.51 ± 0.01	7.50 ± 0.01
	Hypoxia with vitamin C	7.50 ± 0.01	7.52 ± 0.01	7.54 ± 0.01*‡	7.52 ± 0.01	7.52 ± 0.01	7.51 ± 0.01	7.50 ± 0.01	7.51 ± 0.01	7.51 ± 0.01
	Normoxia with vitamin C	7.49 ± 0.01	7.50 ± 0.01	7.50 ± 0.01	7.49 ± 0.01	7.49 ± 0.01	7.51 ± 0.01	7.49 ± 0.01	7.50 ± 0.01	7.49 ± 0.01
PaCO ₂ (mmHg)	Normoxia	32.5 ± 0.7	34.4 ± 0.9	35.8 ± 0.7	35.2 ± 0.8	37.0 ± 1.3	38.0 ± 1.4	36.7 ± 1.3	38.1 ± 1.5	38.1 ± 1.0
	Hypoxia	32.8 ± 0.7	33.7 ± 0.7	$27.9 \pm 0.6^{*} \ddagger$	$28.0 \pm 0.6^{*}$ ‡	$29.6 \pm 0.8^{*} \ddagger$	$29.6 \pm 0.8^{*}$ ‡	$30.4 \pm 1.7^{*} \ddagger$	$30.2 \pm 0.8^{*} \ddagger$	$29.8 \pm 0.9^{*} \ddagger$
	Hypoxia with vitamin C	33 ± 1.0	33.1 ± 0.7	29.4 ± 0.8*‡	29.0 ± 0.8*‡	30.5 ± 1.2*‡	30.2 ± 0.9*‡	30.4 ± 0.8*‡	30.1 ± 0.7*‡	30.9 ± 0.8*‡
	Normoxia with vitamin C	33.3 ± 1.0	35.8 ± 0.8	36.6 ± 0.8	37.1 ± 0.8	36.1 ± 0.6	37.6 ± 0.8	37.6 ± 0.7	37.3 ± 0.9	38.1 ± 0.9
PaO ₂ (mmHg)	Normoxia	103.7 ± 5.1	101.8 ± 3.9	101.0 ± 2.3	107.3 ± 3.1	108.1 ± 1.9	105.9 ± 3.0	109.7 ± 4.8	102.1 ± 2.3	101.7 ± 2.9
	Hypoxia	108.9 ± 3.2	105.6 ± 3.0	$51.7 \pm 2.0^{*} \ddagger$	$48.6 \pm 2.2^{*} \ddagger$	$46.2 \pm 1.6^{*} \ddagger$	$47.0 \pm 1.5^{*} \ddagger$	$46.4 \pm 1.4^{*}$ ‡	$46.8 \pm 1.6^{*} \ddagger$	$45.2 \pm 1.8^{*} \ddagger$
	Hypoxia with vitamin C	108.2 ± 3.9	107.5 ± 3.7	51.1 ± 2.7*‡	47.9 ± 1.8*‡	45.5 ± 1.6*‡	44.8 ± 1.4*‡	44.4 ± 1.5*‡	44.7 ± 1.8*‡	$44.6 \pm 1.8^{*}$ ‡
	Normoxia with vitamin C	109.0 ± 5.0	102 ± 2.7	99.1 ± 2.1	103.4 ± 1.4	105.5 ± 3.4	101.7 ± 2.6	106.6 ± 3.0	106.0 ± 2.3	102.3 ± 2.0
Sat Hb (%)	Normoxia	102.8 ± 1.1	103.1 ± 0.5	103.0 ± 0.5	103.7 ± 0.6	103.7 ± 0.5	104.6 ± 0.5	103.9 ± 0.5	104.4 ± 0.7	103.9 ± 0.6
	Hypoxia	103.3 ± 0.4	101.4 ± 2.5	85.7 ± 2.0* ‡	81.7 ± 2.2*‡	80.2 ± 2.2*‡	$80.3 \pm 2.4^{*}$ ‡	79.3 ± 3.1*‡	81.0 ± 1.9*‡	78.6 ± 2.7*‡
	Hypoxia with vitamin C	102.3 ± 1.2	103.3 ± 0.4	83.4 ± 1.9*‡	80.5 ± 2.3*‡	75.2 ± 2.3*†‡	75.4 ± 2.5*‡	73.2 ± 2.5*†‡	74.3 ± 3.1*†‡	74.0 ± 3.2*†‡
	Normoxia with vitamin C	102.9 ± 0.6	103.4 ± 0.6	102.8 ± 0.7	103.7 ± 0.3	103.6 ± 0.6	104.0 ± 0.5	104.3 ± 0.4	104.5 ± 0.4	103.9 ± 0.5
HCO ₃ ⁻ (mmol/L)	Normoxia	25.2 ± 1.0	26.7 ± 0.8	27.6 ± 0.6	27.0 ± 0.7	27.9 ± 0.8	28.1 ± 0.6	27.2 ± 0.7	28.5 ± 0.7‡	28.9 ± 0.6‡
	Hypoxia	25.4 ± 0.9	25.6 ± 0.7	$23.6 \pm 0.5^{*}$	$22.3 \pm 0.5^{*}$ ‡	23.3 ± 0.6*‡	$23.3 \pm 0.5^{*}$ ‡	$23.8 \pm 1.1^{*}$ ‡	23.6 ± 0.6*‡	$23.7 \pm 0.9^{*}$
	Hypoxia with vitamin C	25.7 ± 1.0	26.7 ± 0.7	$24.8 \pm 0.5^{*}$	23.3 ± 0.6*‡	$23.9 \pm 0.8^{*}$ ‡	23.6 ± 0.4*‡	23.5 ± 0.5*‡	24.0 ± 0.5*‡	$24.0 \pm 0.6^{*}$
	Normoxia with vitamin C	25.8 ± 1.0	27.2 ± 0.7	27.3 ± 0.5	27.8 ± 0.5	27.7 ± 0.7	29.8 ± 0.8‡	28.2 ± 0.7	28.4 ± 0.7‡	28.4 ± 1.0
[Hb] (g/dL)	Normoxia	11.7 ± 0.3	11.2 ± 0.3	11.0 ± 0.3	10.3 ± 0.3‡	9.5 ± 0.3‡	9.9 ± 0.4‡	10.1 ± 0.4 ‡	9.9 ± 0.3‡	9.7 ± 0.3‡
	Hypoxia	12.5 ± 0.3	11.8 ± 0.3	11.9 ± 0.3	$11.5 \pm 0.3^{*}$	$11.7 \pm 0.2^{*}$	$11.9 \pm 0.3^{*}$	$12.2 \pm 0.3^{*}$	$12.5 \pm 0.3^{*}$	$12.7 \pm 0.3^{*}$
	Hypoxia with vitamin C	12.2 ± 0.5	11.9 ± 0.3	12.0 ± 0.2	$11.5 \pm 0.3^{*}$	$11.8 \pm 0.2^{*}$	$12.2 \pm 0.2^{*}$	$12.5 \pm 0.2^{*}$	$12.7 \pm 0.2^{*}$	13.1 ± 0.3*‡
	Normoxia with vitamin C	11.1 ± 0.4	11.1 ± 0.2	10.8 ± 0.2	9.8 ± 0.3‡	9.7 ± 0.3‡	10.0 ± 0.3‡	10.1 ± 0.4‡	10.2 ± 0.4‡	10.3 ± 0.4‡

Abbreviations: dGA, days of gestation; [Hb], hemoglobin concentration; HCO₃⁻, bicarbonate; PaCO₂, arterial blood partial pressure of carbon dioxide; PaO₂, arterial blood partial pressure of oxygen; Sat Hb, arterial blood percentage saturation of hemoglobin with oxygen.

https://doi.org/10.1371/journal.pbio.2006552.t001

dGA 130 and dGA 135 and significant reductions in maternal hemoglobin concentration ([Hb]) from 110 dGA compared to baseline (Table 1). In control ewes, maternal plasma total levels of plasma concentrations of total $NO_3^- + NO_2^-$ (NOx) also showed an increase with advancing gestation, with a significant difference from baseline at 135 dGA (Fig 1E).

Exposure of pregnant sheep to a 10% inspired fraction of oxygen for a month in the last third of gestation, from 105 to 135 dGA, led to a sustained controlled reduction in the maternal

PaO₂ and arterial blood percentage saturation of hemoglobin with oxygen (Sat Hb) (Fig 1B– 1D). Chronic hypoxia in untreated ewes led to a transient maternal respiratory alkalosis, with significant falls in maternal arterial blood partial pressure of carbon dioxide (PaCO₂) throughout exposure and a significant increase in maternal pH at 106 dGA, the day after the onset of hypoxia (Table 1). This maternal respiratory alkalosis was buffered by reductions in maternal HCO_3^- throughout the chronic hypoxia period (Table 1). In contrast to control ewes undergoing normoxic pregnancy, ewes exposed to chronic hypoxia did not show a significant fall from baseline in maternal [Hb] and levels of maternal [Hb] were significantly higher than those in control ewes from 110 dGA (Table 1). Maternal hypoxia in untreated ewes enhanced the ontogenic increase in maternal plasma NOx (Fig 1E), and it did not affect maternal food intake or maternal weight gain (Fig 1F and 1G).

Maternal vitamin C treatment in both control and hypoxic ewes produced similar increments from baseline (N: 38.0 ± 3.9 ; H: $36.2 \pm 3.1 \mu mol/L$) in maternal plasma vitamin C, doubling the circulating concentration (Fig 1H). Maternal vitamin C treatment in hypoxic pregnancy did not affect the alterations in maternal blood gases or pH or maternal plasma NOx seen in untreated hypoxic ewes. In addition, ewes undergoing hypoxia treated with vitamin C also did not show any changes in maternal food intake (Fig 1B–1F and Table 1). However, maternal treatment with vitamin C in hypoxic pregnancy led to a greater fall in maternal Sat Hb despite a similar fall in maternal PaO₂ compared to untreated ewes (Fig 1C and 1D). This meant a rightward shift in the oxygen–hemoglobin dissociation curve and thereby a significant increase in the maternal arterial P₅₀ from 23.5 ± 0.9 to 26.6 ± 10 mmHg (P < 0.05; Fig 11). The total number of placentomes, the placental weight, the fetal:placental weight ratio, and the distribution of placentome type did not differ among all 4 groups (see <u>S1A–S1C Fig</u>).

Fetuses from hypoxic pregnancy showed growth restriction with brain sparing. They had a significant reduction in body weight, body mass index (BMI), and lower limb length (LLL) and significant increases in the absolute and relative brain weight, in the relative weight of the hypothalamus, the absolute weight of the cerebral hemispheres, and in the ratio of the biparie-tal diameter relative to LLL when compared to fetuses from normoxic pregnancy (Table 2 and Fig 3A–3F). Fetuses from hypoxic pregnancy showed an increase in [Hb] and in plasma total NOx (Fig 3G and 3H). They also had increased hepatic nitrotyrosine concentrations and decreases in plasma homocysteine and in the activities of catalase and superoxide dismutase (SOD) in the liver (Fig 3I–3L).

Fetuses from hypoxic pregnancy treated with maternal vitamin C no longer showed significant reductions in body weight and BMI, no longer showed a significant increase in relative brain weight, and the ratio of the biparietal diameter relative to LLL was restored when compared to fetuses from normoxic pregnancy. However, the reduction in LLL and the increase in the relative weight of the hypothalamus persisted (<u>Table 2</u> and <u>Fig 3A-3F</u>). Fetuses from hypoxic pregnancy treated with maternal vitamin C still showed an increase in [Hb] and total plasma NOx and a fall in liver SOD, and the increase in hepatic nitrotyrosine was significantly greater than in fetuses from untreated hypoxic pregnancy. However, they showed restored levels of plasma homocysteine and of hepatic catalase activity when compared to hypoxic fetuses of untreated pregnancy (<u>Table 2</u> and <u>Fig 3J and 3K</u>).

Maternal treatment with vitamin C in normoxic pregnancy had no effect on any outcome variable measured in the ewes or fetuses when compared to untreated normoxic pregnancy (Tables $\underline{1}$ and $\underline{2}$, Figs $\underline{1}$ and $\underline{3}$).

When blood was taken at post mortem immediately following the end of the experiment at 138 dGA, the fetal plasma concentrations of cortisol were not different among all groups (N: 17.6 ± 3.0 ; H: 16.6 ± 2.9 ; HC: 27.7 ± 2.9 ; NC: 28.5 ± 6.2 ng.mL⁻¹). We have also reported that

Table 2. Fetal and adult offspring organ weights. Values are mean \pm SEM for the fetal offspring organ weights at 138 dGA and adult offspring organ weights at PM. Groups are normoxia (fetus, *n* = 12; adult, *n* = 9), hypoxia (fetus, *n* = 6; adult, *n* = 9), hypoxia with vitamin C (fetus and adult, *n* = 9), and normoxia with vitamin C (fetus and adult, *n* = 9). Significant (*P* < 0.05) differences are *versus normoxia, †hypoxia versus hypoxia with vitamin C, and two-way ANOVA with posthoc Tukey test.

	Normoxia	Нурохіа	Hypoxia with vitamin C	Normoxia with vitamin C
Fetal offspring				
Brain				
Absolute (g)	43.3 ± 0.7	38.9 ± 2.3*	41.5 ± 1.0	44.0 ± 1.5
Relative (g/kg)	12.1 ± 0.6	$14.9 \pm 1.0^{*}$	13.0 ± 0.8	13.7 ± 1.1
Hypothalamus				
Absolute (g)	3.7 ± 0.2	4.08 ± 0.2	4.3 ± 0.1	4.2 ± 0.2
Relative to brain	0.086 ± 0.006	$0.106 \pm 0.004^*$	$0.105 \pm 0.004^*$	0.096 ± 0.004
Cerebellum				
Absolute (g)	4.6 ± 0.2	4.0 ± 0.2	4.5 ± 0.1	4.6 ± 0.2
Relative to brain	0.106 ± 0.003	0.104 ± 0.002	0.108 ± 0.003	0.105 ± 0.002
Hemispheres				
Absolute (g)	33.6 ± 0.9	$28.5 \pm 1.8^{*}$	29.8 ± 1.5	32.7 ± 1.5
Relative to brain	0.780 ± 0.016	0.733 ± 0.007	0.717 ± 0.030	0.756 ± 0.010
Heart				
Absolute (g)	33.6 ± 2.5	26.3 ± 3.0	32.2 ± 1.6	33.7 ± 2.2
Relative (g/kg)	9.2 ± 0.6	9.7 ± 0.2	10.0 ± 0.5	10.3 ± 0.7
Liver				
Absolute (g)	78.8 ± 5.2	56.6 ± 7.2	69.5 ± 5.8	67.2 ± 8.4
Relative (g/kg)	21.4 ± 0.8	21.0 ± 1.8	21.1 ± 1.2	18.8 ± 1.0
Adult offspring				
Brain				
Absolute (g)	79.9 ± 1.8	84.6 ± 2.1	82.0 ± 2.1	79.8 ± 2.5
Relative (g/kg)	3.2 ± 0.2	2.8 ± 0.1	3.0 ± 0.2	2.8 ± 0.1
Hypothalamus				
Absolute (g)	8.6 ± 0.3	9.2 ± 0.3	9.2 ± 0.3	8.6 ± 0.5
Relative to brain	0.108 ± 0.003	0.109 ± 0.002	0.112 ± 0.004	0.107 ± 0.004
Cerebellum				
Absolute (g)	10.4 ± 0.3	10.6 ± 0.3	10.2 ± 0.4	10.1 ± 0.4
Relative to brain	0.130 ± 0.003	0.125 ± 0.002	0.124 ± 0.004	0.127 ± 0.002
Hemispheres				
Absolute (g)	60.9 ± 1.4	64.8 ± 1.6	62.7 ± 1.7	61.1 ± 1.7
Relative to brain	0.762 ± 0.004	0.766 ± 0.003	0.764 ± 0.005	0.766 ± 0.005
Heart				
Absolute (g)	221.9 ± 10.9	239.5 ± 10.7	239.4 ± 9.1	226.4 ± 8.5
Relative (g/kg)	8.4 ± 0.3	8.0 ± 0.4	8.6 ± 0.5	8.0 ± 0.3
Liver				
Absolute (g)	397.4 ± 18.9	441.9 ± 29.0	407.6 ± 11.5	457.8 ± 16.8
Relative (g/kg)	15.3 ± 1.2	14.9 ± 1.2	14.7 ± 0.9	16.2 ± 0.8

https://doi.org/10.1371/journal.pbio.2006552.t002

PLOS BIOLOGY

maternal plasma stress hormone mean levels are not different in ewes undergoing normoxic or chronic hypoxic pregnancy [<u>39</u>].

Prenatal hypoxia ± antioxidant treatment: Effects on the adult offspring

A separate cohort of animals was allowed to deliver naturally at term under normoxic conditions, and lambs were maintained with their mothers until weaning and then lived on pasture



Fig 3. Fetal offspring data. Measured at 138 dGA: 3A, fetal body weight; 3B, CRL; 3C, BMI; 3D, LLL; 3E, brain weight relative to body weight; 3F, BPD to LLL ratio; 3G, blood concentration of hemoglobin; 3H, NOx; 3I, liver concentrations of 3-nitrotyrosine; 3J, plasma concentrations of homocysteine; 3K, liver catalase activity; and 3L, liver SOD activity. Values are mean \pm SEM for all data. Groups are N (open symbols, *n* = 7–13), H (grey symbols, *n* = 6–10), HC (red symbols, *n* = 6–9), and NC (blue symbols, *n* = 5–9). Significant (P < 0.05) differences are the following: *versus normoxia, †H versus HC, and two-way ANOVA with posthoc Tukey test. BMI, body mass index; BPD, biparietal diameter; CRL, crown rump length; dGA, days of gestation; H, hypoxia; HC, hypoxia with vitamin C; LLL, lower limb length; N, normoxia; NC, normoxia with vitamin C; NOx, plasma concentrations of total NO₃⁻ + NO₂⁻; SOD, superoxide dismutase.

https://doi.org/10.1371/journal.pbio.2006552.g003

in fields surrounding the Barcroft Centre at the University of Cambridge. Lambs from hypoxic pregnancy were born earlier in gestation (N: 147.4 ± 0.6 versus H: 143.6 ± 0.6 days, P < 0.05) with a reduced birth weight (N: 3.5 ± 0.1 versus H: 2.9 ± 0.1 Kg, P < 0.05). Despite being born lighter, lambs from hypoxic pregnancy showed accelerated postnatal fractional growth rates (N: 25.1 ± 1.2 versus H: 32.4 ± 1.8 mg.d⁻¹, starting weight⁻¹, P < 0.05), such that body weight at 9 months was no longer different (N: 26.5 ± 1.2 versus H: 30.4 ± 1.7 Kg). There were also no differences in the weight of the brain, heart, and liver nor in the weight of components of the brain, such as the hemispheres, hypothalamus, or cerebellum between lambs from normoxic or hypoxic pregnancy at 9 months (Table 2).

In vivo experiments in chronically instrumented 9-month-old lambs revealed that offspring from hypoxic pregnancy were hypertensive, and they showed greater resting values for femoral blood flow and vascular conductance (Fig 4A-4F). Addressing in vivo mechanisms contributing to the hypertension, adult offspring of hypoxic pregnancy showed enhanced in vivo femoral constrictor responses to phenylephrine (PE) and to angiotensin II (AngII). They also showed enhanced in vivo femoral dilator responses to sodium nitroprusside (SNP), with no significant effect on basal femoral vascular resistance of N(ω)-nitro-L-arginine methyl ester



Fig 4. Adult offspring in vivo and in vitro cardiovascular function. Adult offspring in vivo basal: 4A, mean arterial pressure; 4B, systolic pressure; 4C, diastolic pressure; 4D, femoral blood flow; 4E, femoral vascular conductance; and 4F, heart rate. Adult offspring in

vivo stimulated femoral blood flow and femoral vascular conductance responses to 4G, PE; 4H, Ang II; and 4I, SNP. 4J. Femoral vascular resistance response after 10 minutes of L-NAME exposure (change from baseline). 4K. In vitro concentration–response curves of the percent Rmax to methacholine. 4L. The NO-dependent, EDHF-dependent, and PG-dependent components (AAC) of vasorelaxation in second-order femoral resistance arteries isolated from 9-month-old adult offspring. Values are mean \pm SEM for all data. Groups are N (open symbols, n = 6-11), H (black/grey symbols, n = 6-11), HC (red symbols, n = 5-11), and NC (blue symbols, n = 6-15). Significant (P < 0.05) differences are *versus normoxia, \dagger H versus HC, one- or two-way ANOVA with posthoc Tukey test (4A–4F, 4J and 4L), or two-way repeated measures ANOVA with posthoc Tukey test (4G–4I, 4K). AngII, angiotensin II; AAC, area above the curve; EDHF, endothelium derived hyperpolarising factor; H, hypoxia; HC, hypoxia with vitamin C; L-NAME, N(ω)-nitro-L-arginine methyl ester; N, normoxia; NC, normoxia with vitamin C; PE, phenylephrine; percent Rmax, maximal femoral vessel relaxation; PG, prostaglandin; SNP, sodium nitroprusside.

https://doi.org/10.1371/journal.pbio.2006552.g004

(L-NAME) treatment (Fig 4G-4]). To further address mechanisms, third order femoral vessels isolated from adult offspring of hypoxic pregnancy showed reduced relaxation to methacholine because of endothelial dysfunction due to a decrease in NO-independent mechanisms, including prostanoid and endothelium-derived hyperpolarizing factor (EDHF) dilator reactivity (Fig 4K and 4L).

Adult offspring of hypoxic pregnancy treated with maternal vitamin C were no longer hypertensive (Fig 4A-4C); they showed a significant increase in femoral vascular resistance following L-NAME treatment, indicating greater circulating NO bioavailability (Fig 4I), and had constrictor and dilator responses in vivo and ex vivo, which were more similar to adult offspring of control pregnancy (Fig 4G-4L). The weight of organs collected at 9 months was similar in adult offspring from treated and untreated hypoxic pregnancies (Table 2).

Adult offspring of normoxic pregnancy treated with vitamin C showed similar responses in all outcomes relative to offspring of untreated normoxic pregnancy, with the exception of reduced endothelial function in isolated femoral vessels (<u>Table 2</u> and <u>Fig 4L</u>).

Discussion

Any pregnancy in which there is an increase in placental vascular resistance will impact oxygen delivery to the fetus. Therefore, chronic fetal hypoxia is one of the most common complications of human pregnancy and is known to cause fetal growth restriction [13,14]. Recently, we created isobaric chambers able to maintain pregnant sheep for long periods of gestation under highly controlled hypoxic conditions [39,40]. We have used this novel technology to model the short- and longer-term effects of isolated hypoxic pregnancy on the cardiovascular system of the offspring in a species of similar temporal developmental milestones to humans.

Hypoxic pregnancy, fetal growth, and fetal brain sparing

During healthy pregnancy, maternal cardiovascular adaptations are essential to maintain appropriate fetal growth and development as well as the maternal well-being. These adaptations include the expansion of the maternal blood volume, which places the mother in a state of physiological anemia. Following Poiseuille's Law, reductions in maternal [Hb] decrease the maternal blood viscosity, thereby promoting lower systemic vascular resistance and increased blood flow in the uteroplacental circulation to match fetal growth in the last third of pregnancy [41,42]. Therefore, the fall in maternal [Hb] in control ewes in the present study is consistent with healthy maternal adaptations during late gestation. In humans and sheep, exposure to hypoxia during pregnancy, such as that associated with high altitude, leads to a fall in the maternal PaO₂ and PaCO₂ and an increase in maternal [Hb] [42–44]. In human high-altitude pregnancy, there is also an increase in the maternal blood pH and a decrease in HCO₃⁻ aiming to buffer the maternal alkalosis [43,44]. Therefore, the maternal circulatory responses to chronic hypoxia in the present study are consistent with these maternal adaptations to hypoxic pregnancy, and they explain the significantly higher levels of maternal [Hb] and lack of a fall in maternal [Hb] with advancing gestation in the chronically hypoxic relative to the normoxic ewe.

Fetuses of hypoxic pregnancy were not only lighter but also showed a reduced BMI with an increase in relative brain weight. These are robust indices of fetal brain sparing at the expense of redistribution of blood flow away from the fetal peripheral circulation [14]. Regional differences in brain weight changes suggest that the hypothalamus and cerebellum were preserved more so than the hemispheres. Seminal studies in fetal sheep have previously reported a hierarchy in maintaining blood flow to subcortical regions at the expense of blood flow to the brain hemispheres during chronic hypoxia [45]. In a separate study, we have previously reported that a comparative level of maternal hypoxia in sheep, as used in the present study, reduced the fetal PaO₂ from normal baseline to values of $12 \pm 1 \text{ mmHg}$ [40]. This is of significant clinical translation, as this is the level of reduced fetal oxygenation measured by cordocentesis in human pregnancies complicated by significant fetal growth restriction [46].

Hypoxic pregnancy and oxidative stress in the fetal cardiovascular system

Additional data in the present study show that increased oxidative stress in the fetus contributes to the mechanisms mediating the adverse consequences of developmental hypoxia. 3-nitrotyrosine (3-NT) is a product of protein tyrosine nitration following oxidative damage to proteins by peroxynitrie [47]. An increase in oxidative stress will cause an increased demand for the synthesis of endogenous antioxidant enzymes. Therefore, a decrease in the activities of hepatic catalase and SOD in fetuses from hypoxic pregnancy provides further evidence of increased oxidative stress and impaired antioxidant defenses in the hypoxic fetus in the present study.

It is widely recognized that individuals with high-plasma homocysteine (Hcy) levels have an increased risk of developing cardiovascular disease [48]. However, it must also be recognized that Hcy is an important intermediate in the conversion of methionine to cysteine, which is then used in the production of the antioxidant glutathione. During an oxidative challenge, this pathway is up-regulated rapidly [48,49]. However, in situations of low Hcy, there is a limit to how much glutathione can be produced, thereby restricting the body's ability to respond to oxidative stress [49]. Any condition that causes an increase in oxidative stress will promote an increased demand on the liver to produce glutathione, and consequently, low availability of glutathione and homocysteine is also associated with a large range of diseases [49]. In hypoxic pregnancies, oxidative stress will drive Hcy into glutathione synthesis, leading to a fall in plasma Hcy levels, as was observed with the hypoxic fetuses in the present study. These data in the sheep fetus are also in keeping with reports of oxidative stress in the fetal cardiovascular system in rodent models of hypoxic pregnancy [8,28–31,35,36].

In the present study, treatment of hypoxic pregnancies with vitamin C maintained the increase in fetal [Hb], and in fetal total plasma NOx, it augmented the increase in nitrotyrosine in the fetal liver but prevented the fall in the fetal hepatic activities of Hcy and catalase without affecting the reduction in SOD in the fetus of hypoxic relative to normoxic pregnancy. Maintained or augmented increases in fetal [Hb], fetal total plasma NOx, and in nitrotyrosine in the fetal liver in hypoxic pregnancy suggests that maternal treatment with vitamin C is not protective in hypoxic pregnancy by affecting oxygen sensing pathways, the induction of oxidative stress, or compensatory adaptations to increase NO bioavailability in the fetus. Rather, maternal treatment with vitamin C in hypoxic pregnancy confers antioxidant protection by additional exogenous antioxidant supplementation. This normalizes the fetal hepatic activities of

Hcy and catalase but not SOD, thereby reducing the demand for 2 out of 3 fetal endogenous antioxidant defenses in hypoxic pregnancy.

Maternal vitamin C protects fetal growth and prevents programmed systemic hypertension in the adult offspring in hypoxic pregnancy

Maternal vitamin C treatment increases umbilical blood flow in vivo in ovine pregnancy by quenching O_2^- production and increasing NO bioavailability [33,34]. Here, we show that maternal vitamin C treatment additionally increases the maternal P_{50} and restores fetal plasma Hcy levels and hepatic catalase activity. Therefore, maternal vitamin C treatment by increasing umbilical blood flow, fetal endogenous antioxidant defenses, and the transplacental PO₂ gradient will buffer reductions in fetal oxygen delivery despite chronic hypoxia. These mechanisms may explain the capacity of the chronically hypoxic fetus whose mother is treated with vitamin C to be able to maintain appropriate growth, no longer requiring fetal brain sparing. Pregnancy complicated by fetal growth restriction is itself a major killer in perinatal medicine today [15]. Therefore, treatment to protect fetal growth in human high-risk pregnancy is clinically important.

Systemic hypertension affects 1 billion adults worldwide [50]. Increased systolic blood pressure is an independent risk factor for coronary events, stroke, and heart failure [51]. A few studies in rodents have shown that hypoxic pregnancy can program hypertension in the adult offspring and that this was associated with constrictor hyper-reactivity in the peripheral vasculature and exacerbated cardiovascular responses to stress [52-55]. However, intervention against the programmed hypertension was not addressed in these studies. Here, we show that ovine pregnancy complicated by developmental hypoxia can also program significant hypertension in the adult offspring. The mechanisms mediating hypertension in the present study include increased in vivo vasoconstrictor reactivity to α -adrenergic agonists and to AngII in the peripheral circulation. In addition, we show endothelial dysfunction due to impaired EDHF and prostanoid-dependent reactivity in femoral arteries of adult offspring of hypoxic pregnancy. There is also in vivo evidence of programmed compensatory increases in NO function in the cardiovascular system in adult offspring of hypoxic pregnancy. We show in vivo that they have enhanced basal femoral blood flow and vascular conductance and a greater femoral dilator response to SNP, a NO donor. In the present study, maternal vitamin C treatment in hypoxic pregnancy prevented hypertension in the adult offspring. Therefore, we provide novel in vivo evidence of successful intervention with antenatal maternal antioxidant administration against programmed systemic hypertension in the adult offspring. Further, the successful intervention against programmed hypertension is in an ovine model of chronic fetal hypoxia, thereby of comparable developmental milestones to humans. In the present study, maternal treatment with vitamin C conferred protection against hypertension in the offspring by programming an increase in NO bioavailability, since in vivo treatment of adult offspring with L-NAME led to significantly greater increases in femoral vascular resistance. Further, maternal vitamin C also improved endothelial function. Combined, therefore, these protective mechanisms induced by maternal vitamin C treatment meant that NO-compensatory responses in the peripheral vasculature of adult offspring of hypoxic pregnancy, such as enhanced basal femoral blood flow and vascular conductance and a greater femoral dilator response to SNP, were no longer required.

In the present study, fetal sheep were not instrumented with catheters, as the focus was on offspring of hypoxic pregnancy being allowed to be born naturally and maintained until adulthood for cardiovascular study. This study design precluded the measurement of fetal arterial blood pressure and knowledge of whether chronic fetal hypoxia led to fetal arterial hypertension in utero, which was maintained or exacerbated until adulthood. However, in separate studies, which determined changes in fetal arterial blood pressure during hypoxic pregnancy in sheep using isobaric chambers, we [40] and others [56] have reported that the chronically hypoxic fetus showed an impaired or unchanged ontogenic increase in basal arterial blood pressure. Similarly, late-gestation fetal sheep conceived, gestated, and studied at high altitude had lower resting arterial blood pressure relative to sea level controls [57]. Combined, therefore, past and present evidence suggests that chronic fetal hypoxia programs adult-onset hypertension and that this develops in association with aging, rather than it persisting from fetal origin.

Several studies in rodents and in sheep have reported that pregnancy exposed to excess glucocorticoids can also program hypertension in the adult offspring [58–62]. In the present study, fetal plasma glucocorticoid levels were not elevated by the end of hypoxic pregnancy relative to normoxic pregnancy. This is again consistent with data derived from high-altitude pregnancy, which reports in sheep that the fetal hypothalamo–pituitary–adrenal axis desensitizes in response to chronic hypoxia. This is an adaptive response to prevent stress-induced preterm birth and to maintain normal circulating levels of fetal cortisol for appropriate development and maturation [63]. Therefore, in the present study, chronic fetal hypoxia programmed adult-onset hypertension independent of fetal hypercostisolemia.

In the clinical setting, people are diagnosed as hypertensive when the systolic/diastolic reading exceeds 140/90, with systolic blood pressure being the greatest predictor of later disease (National Institute for Health and Care Excellence [NICE] guidelines) [64]. There is no equivalent definition for sheep, and although a direct comparison cannot be made, adult offspring of hypoxic pregnancy in the present study had a mean increase in systolic and diastolic blood pressure values of 13 and of 8 mmHg above control values, respectively. The magnitude of these effects is of high clinical significance, as reports on the prevention and treatment of high blood pressure in humans show that the risk of cardiovascular disease doubles with each increment of systolic/diastolic pressure of 20/10 mmHg and that such individuals require active health-promoting lifestyle modifications to prevent overt cardiovascular disease [18]. The Framingham study reported that modest increases in mean blood pressure milder than those observed in lambs from hypoxic pregnancy in the present study can dramatically increase the risk of a future cardiovascular event [65]. A recent randomized clinical trial also reported that targeting a systolic blood pressure to <120 mmHg compared with <140 mmHg in humans resulted in lower rates of fatal and nonfatal cardiovascular events [66]. In the present study, offspring of hypoxic pregnancy therefore already fall under the human classification of prehypertensive despite being young adult and female and thereby likely to show even greater risk of cardiovascular disease with aging.

Interestingly, adult offspring born from normoxic pregnancy treated with vitamin C also showed evidence of endothelial dysfunction in the present study. While antioxidant supplementation may benefit conditions of increased oxidative stress, antioxidant excess resulting from antioxidant supplementation under normal conditions may paradoxically promote oxidative stress [32]. One mechanism may be by providing excess NO bioavailability, which also serves as a precursor for peroxynitrite generation [32]. This reveals an equally important translational message in the present study, and that is that clinically, maternal treatment with antioxidants should only be administered to pregnancies diagnosed with chronic fetal hypoxia rather than given prophylactically to all pregnancies. This will only be feasible if risk can be identified prior to maternal therapy. Expert obstetric opinion today dictates that in human pregnancy, chronic fetal hypoxia can be reliably identified according to severity by ultrasound scan between 20–26 weeks of gestation. Diagnosis encompasses early onset intrauterine growth restriction, reduced fetal heart rate variability and body movements and/or abnormal

Doppler blood flow velocimetry indices in the fetal middle cerebral artery and the umbilical circulation [67,68]. Although this supports possible translation to the human clinical setting, there are several important points for additional discussion and further key basic science experiments that are required to inform double-blind, randomized, placebo-controlled human clinical trials before maternal antioxidant supplementation in human obstetric practice could be considered.

Limitations

A limitation of the current program of work is that although the study design controlled for sex differences in the offspring, it did not address them. To make the study viable ethically and economically, every singleton pregnancy generated was used. Therefore, studies in the fetal period used the male offspring while studies in the adult period used the female offspring, as ewe lambs are easier to group house when compared to growing rams. Longitudinal comparisons from the fetal through the adult period thus require caution.

A second limitation of the current program of work is that outcome variables in the fetal and adult periods were not balanced for sex. Several recent reviews have reported important sex-dependent differences in the programming effects on cardiovascular function in the adult offspring of adverse intrauterine conditions, usually highlighting that the male offspring is more vulnerable [69–71]. It is also well accepted that aging increases cardiovascular risk [72–74] and that ovarian estrogens at adulthood confer protection against cardiovascular disease [75]. Given that the present study reported significant hypertension in young adult females born from hypoxic pregnancy, the data may underestimate the potential adverse impact of developmental hypoxia on systemic hypertension and cardiovascular dysfunction in adult male offspring in this ovine model of hypoxic pregnancy. However, the study does emphasize that hypoxic pregnancy in sheep increases the risk of cardiovascular dysfunction and that maternal antioxidant treatment is protective in both male and female offspring.

A third limitation of the current study design is that pregnancies exposed to hypoxia from 105 to 135 dGA were returned to normoxia at 135 days to allow delivery under normal oxygen conditions. This study design ensured a fixed length period of fetal exposure to chronic hypoxia prior to birth. However, it also means that adult-onset hypertension could be programmed by fetal exposure to chronic hypoxia with some reoxygenation rather than by chronic fetal hypoxia alone. Since hypoxia and reoxygenation is a more potent stimulus of oxidative stress than hypoxia alone [32], the data highlight the protective impact of maternal antioxidant therapy even under these conditions.

The final limitation and perhaps the most important in terms of human clinical translation is that multicenter human clinical studies have been unable to confirm benefits of maternal treatment with antioxidant vitamins in pregnancies at risk of preeclampsia [38,76]. These studies include the vitamins in preeclampsia (VIP) and international trial of antioxidants in the prevention of preeclampsia (INTAPP) trials, which found that administration of vitamin C (1 g.d⁻¹) and E (400 IU) to mothers once clinical signs of the preeclampsia were established did not affect the incidence of preeclampsia but instead led to an increase in the rate of low-birth weight babies by 4%. While preeclampsia was not the focus of the present study and the adverse effect on birth weight of maternal vitamin C in human studies appears no longer present with meta-analysis [38,76], these clinical findings are significant enough to warn against possible adverse side effects of maternal vitamin C treatment in human pregnancy. Hence, we strongly agree that vitamin C may not be the antioxidant of choice for translation to human therapy. However, the data we show provide proof-of-principle that maternal treatment with antioxidants can protect against developmental programming of adult-onset hypertension. Future key experiments to inform human clinical trials should consider alternative maternal antioxidant therapy of improved translational value using human clinically relevant animal models of adverse pregnancy, such as sheep. Plausible alternative candidate antioxidant therapies for human translation may include melatonin, allopurinol, or the mitochondria-targeted antioxidant MitoQ. Rodent models show that maternal treatment with melatonin confers protection on fetal growth and cardiovascular function in the offspring of adverse pregnancy at doses similar to or lower than those required to avoid jet lag in humans [77,78]. An alternative antioxidant strategy may be to prevent the synthesis of free radicals through specific pathways rather than quench them once formed, as in the case of the xanthine oxidase inhibitor allopurinol. Hypoxia is a potent stimulus for xanthine oxidase-induced superoxide anion generation [32], and recently, we have reported that maternal treatment with allopurinol in hypoxic pregnancy in rats protects against cardiac dysfunction in the adult offspring [79]. Yet another strategy may be for targeted antioxidant therapy. Mitochondria are a major site of ROS production; therefore, targeting them specifically should be one of the most effective antioxidant therapeutic strategies. This is now possible with mitochondria-targeted antioxidants. MitoQ is composed of a lipophilic triphenylphosphonium cation covalently attached to an ubiquinol antioxidant [80,81]. Lipophilic cations can easily move through phospholipid bilayers without requiring a specific uptake mechanism. Therefore, the triphenylphosphonium cation concentrates MitoQ several hundred-fold within the mitochondria, driven by the large mitochondrial membrane potential [78,79]. Only within the mitochondria, MitoQ is reduced by the respiratory chain to its active ubiquinol form, which is a particularly effective antioxidant that prevents lipid peroxidation and mitochondrial damage [80,81]. The benefits of MitoQ have been revealed in a range of in vivo studies in rats and mice and in two human trials [82-85]. In contrast to vitamin C and other conventional antioxidants, MitoQ demonstrates no pro-oxidant activity at high doses and long-term administration to mice for 28 weeks [83] and to human patients in two Phase II trials, one that lasted 1 year revealed no toxicity [84,85]. Two recent studies in rats have also reported that nano-particle bound MitoQ, which prevents passage of the drug through the placenta and to the fetus, protected against programmed cardiac diastolic dysfunction, endothelial dysfunction, and neurodegeneration in the adult offspring, using an established model of hypoxic pregnancy in rats [35,86]. However, the antioxidant benefits of either melatonin or allopurinol or MitoQ in protecting against fetal growth restriction and programmed hypertension at adulthood in offspring of high-risk pregnancy in sheep have yet to be determined.

In conclusion, our discoveries provide compelling evidence of clinical translational importance for chronic fetal hypoxia programming of adult-onset hypertension in a species of similar developmental milestones to humans. Further, the data provide novel insight into underlying mechanisms and thereby successful intervention against cardiovascular dysfunction in the next generation programmed by pregnancy complicated by developmental hypoxia (See Summary illustration, Fig.2).

Materials and methods

Ethics statement

All procedures were performed under the Home Office Project Licence PC6CEFE59 under the Animals (Scientific Procedures) Act 1986 Amendment Regulations 2012, following ethical review by the University of Cambridge Animal Welfare and Ethical Review Board (AWERB).

Surgical preparation of pregnant ewes

At 100 ± 1 days gestational age (term ca. 145 days), pregnant Welsh Mountain ewes carrying singleton pregnancies determined by ultrasound scan (Toshiba Medical Systems Europe,

Zoetermeer, the Netherlands) underwent a laparotomy and catheterization, as previously described [33,34,39,40]. In brief, under general anesthesia (1.5%–2.0% isofluorane in 60:40 O₂: N₂O) maintained by use of a positive pressure ventilator (Datex-Ohmeda Ltd, Hatfield, Hertfordshire, UK), a midline abdominal incision and uterotomy was used to expose the fetal hind limbs and determine fetal sex. If male, then the fetuses were assigned to the fetal studies group; female fetuses were assigned to the adult offspring studies group. The fetus was returned into the intrauterine cavity, and the uterine and maternal abdominal incisions were closed in layers. A Teflon catheter (i.d. 1.0 mm, o.d. 1.6 mm, Altec, UK) was placed in the maternal femoral artery and positioned in the descending aorta, in addition to a maternal femoral venous catheter into the inferior vena cava (Critchly Electrical Products, NSW, Australia). Catheters were filled with heparinized saline (80 i.u.ml⁻¹ heparin in 0.9% NaCl), tunnelled subcutaneously, and exteriorized via a keyhole incision made in the maternal flank to be kept inside a plastic pouch sewn onto the maternal skin.

Following surgery, ewes were housed in individual floor pens with a 12:12-hour light–dark cycle. From 103 dGA, ewes were fed daily a bespoke maintenance diet made up of concentrate and hay pellets to facilitate the monitoring of food intake (Cambridge ewe diet: 40g nuts/kg and 3g hay/kg; Manor Farm Feeds Ltd; Oakham, Leicestershire, UK) [39,40]. On day 105 of gestation, ewes were randomly assigned to 1 of 4 experimental groups: normoxia (N), chronic hypoxia (H), chronic hypoxia with vitamin C treatment (HC), and normoxia with vitamin C (NC); n = 18 for all groups.

Chronic hypoxia protocol

Ewes were housed in bespoke isobaric hypoxic chambers (Telstar Ace, Dewsbury, West Yorkshire, UK) supplied with controlled volumes of nitrogen and air provided by nitrogen generators and air compressors, respectively, from a specially designed nitrogen generating system (Domnick Hunter Gas Generation, Gateshead, Tyne & Wear, UK) [39,40]. Ambient PO₂, PCO₂, humidity, and temperature within each chamber were monitored via sensors, displayed, and values recorded continuously via the Trends Building Management System of the University of Cambridge through a secure Redcare intranet. In this way, the percentage of oxygen in the isolators could be controlled with precision over long periods of time. For experimental procedures, each chamber had a double transfer port to internalize material and a manually operated sliding panel to bring the ewe into a position in which daily sampling of blood could be achieved through glove compartments. Each chamber incorporated a drinking bowl on continuous water supply and a rotating food compartment, which could be removed for determining food intake. Therefore, all experimental and maintenance procedures could be carried out without interruption of the hypoxic exposure. Pregnancies randomly assigned to the chronic hypoxia group were placed inside the chambers at 103 dGA under normoxic conditions (11 L.sec⁻¹ air, equating to $39.6 \text{ m}^3.\text{h}^{-1}$). At 105 days, pregnancies were exposed to ca. 10% O₂ by altering the inspirate mixture to 5 $L.sec^{-1}$ air: 6 $L.sec^{-1}$ N₂. The inspirate air mixture underwent a minimum of 12 changes per hour in each chamber, and the incoming air mixture was passed via silencers able to reduce noise levels within the hypoxic chamber laboratory (76 dB[A]) and inside each chamber (63 dB[A]) to values lower than those necessary to abide by the Control of Noise at Work Regulations. This not only complied with human health and safety and animal welfare regulations but also provided a highly tranquil environment for the animal inside each chamber.

Maternal vitamin C treatment

Starting on day 105 of gestation, vitamin C (Ascorbate; A-5960; Sigma Chemicals, UK; 1.14 mmol/kg/day dissolved in 0.6 ml/kg saline and administered as a slow IV bolus injection) or

saline vehicle (0.6 ml/kg slow IV bolus injection) were administered every day to the mothers at approximately 09:00. The dose of vitamin C used in this study was derived from previous studies in sheep pregnancy in our laboratory, which achieved elevations in circulating ascorbate concentrations within the required range for vitamin C to compete effectively in vivo with NO in ovine pregnancy [33,34,37].

Blood sampling regimen and analysis

Samples of descending aortic maternal blood (0.3 ml) were taken daily for measurement of maternal blood gases and pH and hemoglobin concentration, as previously described [33,34,39,40]. Chamber oxygenation and maternal blood gases and pH were recorded at 104, 105, and 106 dGA then as summary averages of the preceding 5 days for 110, 115, 120, 125, 130, and 135 days. For all data, baseline for each animal was calculated as the average values of 104 and 105 days. On days 104, 105, 106, and 110 of gestation, and every 5 days after that, an additional 9 ml of maternal blood was taken and divided into tubes containing the anticoagulant EDTA or heparin. Samples were spun in a centrifuge for 5 minutes at 1000 x g and 4°C, after which the plasma was aliquoted into storage tubes and immediately frozen for subsequent analysis.

Post mortem of ewes with fetuses

At 138 dGA, pregant ewes with male fetuses were transferred from the hypoxic chambers to the post mortem laboratory wearing a respiratory hood providing the same hypoxic mixture. Under hypoxic conditions, ewes and their fetuses were humanely killed by overdose of sodium pentobarbitone (0.4 ml.kg^{-1} IV Pentoject; Animal Ltd, York, UK) and the fetus exteriorized by Cesarean section. An 8-ml sample of fetal blood was taken from the umbilical artery using a syringe with an attached needle, divided into tubes containing the anticoagulant EDTA or heparin, then spun in a centrifuge for 5 minutes at 1000 x g and 4°C. The plasma was aliquoted into storage tubes and immediately frozen for subsequent analysis. Fetal crown rump length, biparietal diameter, and hind limb lengths were then determined. The upper hind limb comprised the length of the femur, the middle hind limb ranged between the patella and tuber calcis, and the lower hind limb ranged between the tuber calcis and the tips of the phalanges. Remaining fetal organs were dissected, weighed, and immediately frozen for subsequent analysis.

Maternal and fetal blood and plasma analysis

The maternal arterial blood P_{50} , the PO₂ at which 50% of the maternal hemoglobin is saturated with oxygen, was calculated using the average of the maternal values for PO₂ and Sat Hb from day 115–135 of gestation, according to the Hill equation [87]. Maternal and fetal plasma concentrations of NOx species (NO₂⁻ and NO₃⁻) were determined by a commercially available assay kit (Caymen Chemical, USA, Cat No. 780001) according to the manufacturer's instructions. A standard curve was plotted in Excel using a straight line fit, allowing NOx concentrations to be calculated for each sample. The inter- and intra-assay coefficients of variation were 3.4% and 2.7%, respectively, and the lower limit of detection of the assay was 2.5 µmol/L. Maternal plasma concentrations of ascorbic acid were measured by a fluorimetric technique using a centrifugal analyzer with a fluorescence attachment, according to the method of Vuilleumier and Keck [88], in collaboration with the Core Biochemical Assay Laboratory, Cambridge, UK. The interassay coefficients of variation were 7.9% at 27.1 µmol/L and 5.0% at 89.7 µmol/L. The lower limit of detection of the assay was 10 µmol/L. Fetal plasma concentrations of total L-homocysteine were measured using a commercially available enzyme immunoassay kit (Axis-Shield diagnostics Ltd., UK, Cat No. FHCY100), in collaboration with the Core Biochemical Assay Laboratory, Cambridge, UK. The inter- and intra-assay coefficients of variation for a 6.1 µmol/L sample were 2% and 8%, respectively, and the lower limit of detection of the assay was 1.0 µmol/L. Fetal plasma cortisol concentrations were measured using a commercially available ELISA kit (IBL international, Germany, Cat No. RE52061), according to the manufacturer's instructions. The lower limit of detection of the assay was 2.46 ng/mL. The cross-reactivity of the antiserum with other cortisol-related compounds was 4.2% cortisone, 1.4% corticosterone, 0.4% progesterone, and 7.0% deoxycortisol.

Fetal liver oxidative stress analysis

The expression of 3-NT and SOD and the activity of catalase in frozen, powdered, fetal right liver lobe samples were determined by commercial assay kits (3-NT: ab116691, Abcam, Cambridge, UK, AMS biotechnology, Abington, UK, SOD: Sigma-Aldrich, Catalase: 707002), according to the manufacturer's instructions.

Placental analysis

Placentomes were classified into 4 categories by their gross morphological appearance, according to Vatnick and colleagues [89]. Following classification, the individual types were counted and weighed.

Postnatal care of offspring

At 138 dGA, pregnant ewes carrying female fetuses were transferred from the hypoxic chambers to individual pens in a barn with a 12:12-hour light–dark cycle, where they were returned to normoxic conditions. Ewes were allowed to deliver naturally and remained with their offspring until weaning. Newborn lambs were weighed within 12 hours of birth and then daily for the first week of life. Thereafter, body weight was recorded at 2 weeks of age, 30 days of age, and then at subsequent 30-day intervals until they were 9 months old.

Adult offspring surgery

At 265 ± 5 days (9 months) of age, when sheep are sexually mature and classified as young adults, the female offspring were surgically instrumented under general anesthesia with vascular catheters and a femoral artery flow probe. In brief, food but not water was withdrawn 10-15 hours before surgery in order to minimize the risk of bloat and regurgitation of the reticulo-rumen contents. On the day of surgery, anesthesia was induced by injection of Alfaxan (1.5-2.5 ml/kg IV alfaxalone; Jurox Ltd., Worcestershire, UK) into the jugular vein and maintained by spontaneous inhalation of 1.5% isoflurane in 60:40 O₂:N₂O (2 L/min; IsoFlo; Abbott laboratories Ltd., Berkshire, UK). Surgery was performed under aseptic conditions. An incision, approximately 3.5 cm long, was made in the medial surface of each of the hind limbs in order to expose the femoral artery and vein, which were catheterized as before. On the contralateral hind limb, a 4SB Transonic flow probe (Transonic Systems Inc., Ithaca, New York, USA) was placed around the main femoral artery for measurement of femoral blood flow. Catheters and the flow probe lead were then tunneled subcutaneously on their respective sides of the body and exteriorized via keyhole incisions made in the animal's flanks. All skin incisions were then sutured closed and plastic pouches were sewn onto each flank to house the exteriorized catheters and flow probe. Following surgery, lambs were housed in individual floor pens with a 12:12-hour light-dark cycle with free access to hay and water. Antibiotics (30 ml/kg IM procaine benzylpenicillin; Depocillin; Intervet UK Ltd., Milton Keynes, UK) were

administered daily to the lamb for 5 days following surgery, and catheters were flushed daily with heparinised saline (100 i.u/ml heparin, 0.9% NaCl). Following at least 5 days of postoperative recovery, basal and stimulated cardiovascular function was assessed in vivo. Post mortem, third-order femoral arteries (internal diameter $< 300 \mu$ m) were isolated and reactivity was determined via in vitro wire myography.

Adult offspring basal and stimulated in vivo cardiovascular function

Experimental protocols were performed following at least 5 days of postoperative recovery. On the morning of an experiment, ewe lambs were moved into a metabolic crate where their arterial catheter was connected to a sterile pressure transducer (Argon Division, Maxxim Medical, Athens, Texas, USA) and their flow probe to a Transonic flow meter (T206; Transonics Systems Inc., Ithaca, NY, USA). On each experimental day, lambs were allowed to acclimatize for 2–3 hours before commencing recordings. Basal cardiovascular variables were recorded continuously for 4–6 hours. A custom-built Data Acquisition System (Maastricht-Programmable AcQuisition system, M-PAQ and IDEEQ software, Maastricht Instruments, the Netherlands; 1000-Hz sample rate) was used to record arterial blood pressure. Blood flow signals from the Transonic flow meter were also directly recorded by the IDEEQ software, and heart rate was calculated continuously online by the program using the femoral blood flow or systolic pulse as a trigger.

On a separate day, stimulated cardiovascular function in the ewe lambs was determined by measuring the change in mean arterial pressure, heart rate, and femoral blood flow in response to increasing bolus doses of the vasoconstrictors PE (0.25, 0.5, 2, 4, 8, 16, 32, and 64 μ g/kg IA, diluted in 1 ml dH2O; L-phenylephrine; P-6126; Sigma Chemicals, UK); and Ang II (5, 10, 20, 40, 80, 160, and 320 ng/kg IA, diluted in 1 ml dH₂O; MP Biomedicals, California, USA) or in response to increasing doses of the NO donor SNP (0.625, 1.25, 2.5, 5, and 10 μ g/kg IA, diluted in 1 ml dH2O; S-0501; Sigma Chemicals, UK). Suitable dose ranges were derived from pilot experiments and previous studies in the literature [90,91]. All doses were administered in a random order. On a separate day, the change in femoral vascular resistance was also determined after intravenous treatment with the NO synthase inhibitor L-NAME (100 mg/kg; Cayman Chemicals, Cambridge, UK).

Post mortem of adult offspring

At the end of all experimental procedures, ewe lambs underwent euthanasia with a lethal overdose of sodium pentobarbital administered to the indwelling venous catheter (200 ml/kg IV Pentoject; Animalcare Ltd., York, UK). The brain, heart, and liver were dissected and weighed. The brain was further dissected into the midbrain plus diencephalon, cerebellum, and right and left hemispheres for compartmental weighing and storage of each section.

In vitro wire myography of adult offspring femoral arteries

Third-order femoral arteries (internal diameter $< 300\mu$ m) were isolated and placed in physiologic buffer solution (PBS). A segment of approximately 2 mm in length was cut and threaded with two 40-µm diameter stainless steel wires. The vessel segment was then mounted on a wire myograph (Multi Wire Myograph System 610M; DMT, Denmark), while bathed in Krebs solution (mM: NaCl 118.5, KCl 4.75, MgSO₄ 7, H₂O 1.2, KH₂PO₄ 1.2, NaHCO₃ 25.0, CaCl₂ 2.5, and glucose 5.5; Sigma) and constantly exposed to a gas mixture of 5% CO₂ and 95% O₂ at 37°C in the myograph chamber. Following a 30-minute equilibration period, the vessel was stretched in a stepwise manner to a standardized tension equivalent to physiologic transmural pressure. Following a 20-minute equilibration period, all vessels were then precontracted with PE (10⁻⁵ M) before assessing endothelium-dependent vasodilator responses to different concentrations of methacholine (MetCh; 10^{-9} to 10^{-4} M; Sigma Aldrich). To determine the relative contribution of endogenous NO, EDHF, and prostanoid to endothelium-dependent relaxation, concentration-response curves to MetCh were also generated following incubation for 10 minutes with L-NAME (10^{-5} M; Sigma Aldrich) and L-NAME plus indomethacin (10^{-6} M; Sigma Aldrich), as previously reported [92]. Vessels were washed repeatedly with Krebs solution and allowed to equilibrate for at least 20 minutes between different concentration-response curves.

Data analysis and interpretation

For the in vivo cardiovascular experiments at adulthood, variables representing basal cardiovascular function represent the average of the 4–6-hour recording period, which was always during the same time of the day. Femoral vascular resistance (FVR) and femoral vascular conductance (FVC) were calculated by applying Ohm's Law to the circulation, using the following equations, for which ABP is the arterial blood pressure and FBF is femoral blood flow: FVR = ABP/FBF and FVC = FBF/ABP. An increase in vascular resistance signifies a reduction in blood flow greater than can be accounted by a reduction in arterial blood pressure, therefore active vasoconstriction. An increase in vascular conductance signifies an increase in blood flow greater than can be accounted by an increase in arterial blood pressure, therefore active vasodilatation.

For the in vivo dose-response experiments at adulthood, maximal changes from baseline in cardiovascular variables were recorded for each dose. The baseline was taken as the preceding 1–2 minutes of stable recording before each dose was given. After each dose, cardiovascular variables were allowed to return to baseline and remain stable for at least 2 minutes before preparing for the next dose.

For the in vitro wire myography experiments at adulthood, femoral arterial responses were analysed using Prism (v.5.0, GraphPad software). Concentration-response curves were analysed using a sigmoidal fit curve. The maximal vessel relaxation (percent Rmax) was expressed as percentage of the contraction induced by PE. The contribution of NO-dependent mechanisms to the relaxation induced by MetCh was calculated by subtracting the area above the curve (AAC) for MetCh-the AAC for MetCh + L-NAME. The contribution of NO-independent mechanisms was the AAC for MetCh + L-NAME. The contribution of prostanoid to the relaxation induced by MetCh was calculated as the AAC for MetCh + L-NAME-the AAC for MetCh + L-NAME have the AAC for MetC

For all in vivo and ex vivo experiments, data are expressed as the mean \pm SEM. All variables were assessed as appropriate either using two-way ANOVA comparing the interactions between oxygenation and treatment or two-way ANOVA with repeated measures comparing the effects of group and dose or time. Where a significant effect was indicated, Tukey's posthoc test was used to isolate the statistical differences (Sigma-Stat 3.5; Chicago, IL, USA and Graph-Pad Prism 6). For all comparisons, statistical significance was accepted when P < 0.05.

Supporting information

S1 Data. Raw data. Complying with data policy, Excel spreadsheet files are provided with the underlying numerical data points for all graphs contained within the manuscript. (XLSX)

S1 Fig. Placental measurements. Measurements at 138 dGA: A, total placentome weight; B, placentome distribution; and C, fetal:placentome weight ratio. Values are mean \pm SEM. Groups are N (open symbols, *n* = 12), H (grey symbols, *n* = 8), HC (red symbols, *n* = 9), and

NC (blue symbols, n = 9). There are no significant differences between groups. dGA, days of gestation; H, hypoxia; HC, hypoxia with vitamin C, N, normoxia; NC, normoxia with vitamin C.

(TIF)

Acknowledgments

We are extremely grateful to the staff of the University of Cambridge Biological Services for helping with the maintenance of the animals at the Barcroft Centre.

Author Contributions

- **Conceptualization:** Kirsty L. Brain, Beth J. Allison, Emilio A. Herrera, Kimberley J. Botting, Dino A. Giussani.
- Formal analysis: Kirsty L. Brain, Dino A. Giussani.
- Funding acquisition: Dino A. Giussani.
- **Investigation:** Kirsty L. Brain, Beth J. Allison, Youguo Niu, Christine M. Cross, Nozomi Itani, Andrew D. Kane, Emilio A. Herrera, Katie L. Skeffington, Kimberley J. Botting, Dino A. Giussani.
- Methodology: Kirsty L. Brain, Beth J. Allison, Youguo Niu, Christine M. Cross, Nozomi Itani, Andrew D. Kane, Emilio A. Herrera, Katie L. Skeffington, Kimberley J. Botting, Dino A. Giussani.
- Project administration: Kirsty L. Brain, Beth J. Allison, Youguo Niu, Christine M. Cross, Nozomi Itani, Andrew D. Kane, Emilio A. Herrera, Kimberley J. Botting, Dino A. Giussani.
- Resources: Katie L. Skeffington, Dino A. Giussani.
- Software: Kirsty L. Brain, Beth J. Allison, Youguo Niu, Christine M. Cross, Nozomi Itani, Andrew D. Kane, Emilio A. Herrera, Katie L. Skeffington, Dino A. Giussani.
- Supervision: Kirsty L. Brain, Beth J. Allison, Youguo Niu, Christine M. Cross, Nozomi Itani, Andrew D. Kane, Emilio A. Herrera, Katie L. Skeffington, Kimberley J. Botting, Dino A. Giussani.
- Validation: Christine M. Cross, Dino A. Giussani.
- Visualization: Kirsty L. Brain, Beth J. Allison, Youguo Niu, Andrew D. Kane.
- Writing original draft: Kirsty L. Brain, Beth J. Allison, Youguo Niu, Nozomi Itani, Andrew D. Kane, Emilio A. Herrera, Katie L. Skeffington, Kimberley J. Botting, Dino A. Giussani.
- Writing review & editing: Kirsty L. Brain, Beth J. Allison, Youguo Niu, Katie L. Skeffington, Dino A. Giussani.

References

- 1. <u>https://www.bhf.org.uk/research/heart-statistics</u>. British Heart Foundation Statistics. Citation accessed March 12 2018.
- Conference Board of Canada. The Canadian heart health strategy: risk factors and future cost implications. Report 2010. (February 2010.
- European Heart Network and European Society of Cardiology. European Cardiovascular Disease Statistics. European Heart Network and European Society of Cardiology: Brussels, Belgium. 2012; ISBN 978-2-9537898-1-2.

- 4. World Health Statistics 2012. World Health Organization. ISBN 9789241564441.
- Nabel EG. Cardiovascular disease. N Engl J Med. 2003; 349: 60–72. <u>https://doi.org/10.1056/ NEJMra035098</u> PMID: <u>12840094</u>
- Gluckman PD, Hanson MA, Cooper C, Thornburg KL. Effect of in utero and early-life conditions on adult health and disease. N Engl J Med. 2008; 359(1): 61–73. <u>https://doi.org/10.1056/NEJMra0708473</u> PMID: 18596274
- 7. Barker DJP. Mothers, babies and disease in later life. London: BMJ Publishing Group. 1994.
- Giussani DA, Davidge ST. Developmental programming of cardiovascular disease by prenatal hypoxia. J Dev Orig Health Dis. 2013; 4(5): 328–37. Review. <u>https://doi.org/10.1017/S204017441300010X</u> PMID: <u>24970726</u>
- Dabelea D, Hanson RL, Lindsay RS, Pettitt DJ, Imperatore G, Gabir MM, Roumain J, Bennett PH & Knowler WC. Intrauterine exposure to diabetes conveys risk for type 2 diabetes and obesity: a study of discordant sibships. Diabetes 2000; 49: 2208–11. PMID: <u>11118027</u>
- Kral JG, Biron S, Simard S, Hould FS, Lebel S, Marceau S & Marceau P. Large maternal weight loss from obesity surgery prevents transmission of obesity to children who were followed for 2 to 18 years. Pediatrics 2006; 118: e1644–9. <u>https://doi.org/10.1542/peds.2006-1379</u> PMID: <u>17142494</u>
- Smith J, Cianflone K, Biron S, Hould FS, Lebel S, Marceau S, Lescelleur O, Biertho L, Simard S, Kral JG & Marceau P. Effects of maternal surgical weight loss in mothers on intergenerational transmission of obesity. J Clin Endocrinol Metab. 2009; 94: 4275–83. <u>https://doi.org/10.1210/jc.2009-0709</u> PMID: 19820018
- Guénard F, Deshaies Y, Cianflone K, Kral JG, Marceau P & Vohl MC. Differential methylation in glucoregulatory genes of offspring born before vs. after maternal gastrointestinal bypass surgery. Proc Natl Acad Sci USA. 2013; 110 (28): 11439–44. https://doi.org/10.1073/pnas.1216959110 PMID: 23716672
- Hutter D, Kingdom J, Jaeggi E. Causes and mechanisms of intrauterine hypoxia and its impact on the fetal cardiovascular system: a review. Int J Pediatr. 2010; 2010: 401323. <u>https://doi.org/10.1155/2010/</u> 401323 PMID: 20981293
- Giussani DA. The fetal brain sparing response to hypoxia: physiological mechanisms. Review. J Physiol. 2016; 594(5): 1215–30. https://doi.org/10.1113/JP271099 PMID: 26496004
- McCormick MC. The contribution of low birth weight to infant mortality and childhood morbidity. N Engl J Med. 1985; 312(2): 82–90. <u>https://doi.org/10.1056/NEJM198501103120204</u> PMID: <u>3880598</u>
- Leeson CP1, Kattenhorn M, Morley R, Lucas A, Deanfield JE. Impact of low birth weight and cardiovascular risk factors on endothelial function in early adult life. Circulation 2001; 103(9): 1264–8. PMID: 11238271
- 17. Barker DJ, Osmond C. Low birth weight and hypertension. BMJ. 1998: 297(6641): 134-5.
- Chobanian AV, Bakris GL, Black HR, Cushman WC, Green LA, Izzo JL Jr., Jones DW, Materson BJ, Oparil S, Wright JT Jr., Roccella EJ, Joint National Committee on Prevention DE, Treatment of High Blood Pressure. National Heart L, Blood I & National High Blood Pressure Education Program Coordinating C. Seventh report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood Pressure. Hypertension. 2003; 42: 1206–1252. <u>https://doi.org/10.1161/01. HYP.0000107251.49515.c2</u> PMID: <u>14656957</u>
- Eriksson JG, Forsen T, Tuomilehto J, Winter PD, Osmond C & Barker DJ. Catch-up growth in childhood and death from coronary heart disease: longitudinal study. BMJ. 1999: 318: 427–431. PMID: <u>9974455</u>
- Nawathe A, David AL. Prophylaxis and treatment of foetal growth restriction. Best Pract Res Clin Obstet Gynaecol. 2018; 49: 66–78 (2018). <u>https://doi.org/10.1016/j.bpobgyn.2018.02.007</u> PMID: <u>29656983</u>
- 21. Fowden AL, Giussani DA, Forhead AJ. Intrauterine programming of physiological systems: causes and consequences. Physiology (Bethesda). 2006; 21: 29–37.
- 22. Zambrano E, Nathanielsz PW. Mechanisms by which maternal obesity programs offspring for obesity: evidence from animal studies. Nutr Rev. 2013; 71 (1): S42–54.
- 23. Segovia SA, Vickers MH, Gray C, Reynolds CM. Maternal obesity, inflammation, and developmental programming. Biomed Res Int. 2014; 2014: 418975. <u>https://doi.org/10.1155/2014/418975</u> PMID: 24967364
- Taylor PD, Samuelsson AM, Poston L. Maternal obesity and the developmental programming of hypertension: a role for leptin. Acta Physiol (Oxf). 2014; 210(3): 508–23 (2014).
- Cottrell EC, Seckl JR, Holmes MC, Wyrwoll CS. Foetal and placental 11β-HSD2: a hub for developmental programming. Acta Physiol (Oxf). 2014; 210(2): 288–95.
- Martinez SR, Gay MS, Zhang L. Epigenetic mechanisms in heart development and disease. Drug Discov Today. 2015; 20(7): 799–811. <u>https://doi.org/10.1016/j.drudis.2014.12.018</u> PMID: 25572405

- Nicholas LM, Morrison JL, Rattanatray L, Zhang S, Ozanne SE, McMillen IC. The early origins of obesity and insulin resistance: timing, programming and mechanisms. Int J Obes (Lond). 2016; 40(2): 229–38.
- Rueda-Clausen CF, Morton JS, Davidge ST. The early origins of cardiovascular health and disease: who, when, and how. Semin Reprod Med. 2011; 29(3): 197–210. <u>https://doi.org/10.1055/s-0031-1275520 PMID: 21710396</u>
- Patterson AJ, Xiao D, Xiong F, Dixon B, Zhang L. Hypoxia-derived oxidative stress mediates epigenetic repression of PKCε gene in foetal rat hearts. Cardiovasc Res. 2012; 93(2): 302–10. <u>https://doi.org/10. 1093/cvr/cvr322</u> PMID: 22139554
- Thompson LP, Al-Hasan Y. Impact of oxidative stress in fetal programming. J Pregnancy 2012; 2012: 582748. <u>https://doi.org/10.1155/2012/582748</u> PMID: <u>22848830</u>
- Giussani DA, Camm EJ, Niu Y, Richter HG, Blanco CE, Gottschalk R, Blake EZ, Horder KA, Thakor AS, Hansell JA, Kane AD, Wooding FB, Cross CM, Herrera EA. Developmental programming of cardiovascular dysfunction by prenatal hypoxia and oxidative stress. PLoS ONE. 2012; 7(2): e31017. <u>https:// doi.org/10.1371/journal.pone.0031017</u> PMID: <u>22348036</u>
- 32. Halliwell B, Gutteridge JMC. Free Radicals in Biology and Medicine. Oxford University Press. 2004.
- Thakor AS, Richter HG, Kane AD, Dunster C, Kelly FJ, Poston L & Giussani DA. Redox modulation of the fetal cardiovascular defence to hypoxaemia. J Physiol. 2010; 588: 4235–4247. <u>https://doi.org/10. 1113/jphysiol.2010.196402 PMID: 20807788</u>
- Thakor AS, Herrera EA, Seron-Ferre M & Giussani DA. Melatonin and vitamin C increase umbilical blood flow via nitric oxide-dependent mechanisms. J Pineal Res. 2010; 49: 399–406. <u>https://doi.org/10. 1111/j.1600-079X.2010.00813.x PMID: 20958954</u>
- Aljunaidy MM, Morton JS, Kirschenman R, Phillips T, Case CP, Cooke CM, Davidge ST. Maternal treatment with a placental-targeted antioxidant (MitoQ) impacts offspring cardiovascular function in a rat model of prenatal hypoxia. Pharmacol Res. 2018; 134: 332–342. <u>https://doi.org/10.1016/j.phrs.2018</u>. 05.006 PMID: 29778808
- Kane AD, Herrera EA, Camm EJ, Giussani DA. Vitamin C prevents intrauterine programming of in vivo cardiovascular dysfunction in the rat. Circ J. 2013; 77(10):2604–11. PMID: <u>23856654</u>
- Jackson TS, Xu A, Vita JA & Keaney JF Jr. Ascorbate prevents the interaction of superoxide and nitric oxide only at very high physiological concentrations. Circ Res. 1998; 83: 916–922. PMID: <u>9797340</u>
- Poston L, Briley AL, Seed PT, Kelly FJ, Shennan AH & Vitamins in Pre-eclampsia Trial C. Vitamin C and vitamin E in pregnant women at risk for pre-eclampsia (VIP trial): randomised placebo-controlled trial. Lancet. 2006; 367: 1145–1154. <u>https://doi.org/10.1016/S0140-6736(06)68433-X</u> PMID: <u>16616557</u>
- Brain KL, Allison BJ, Niu Y, Cross CM, Itani N, Kane AD, Herrera EA, Giussani DA. Induction of controlled hypoxic pregnancy in large mammalian species. Physiol Rep. 2015; 3(12): pii: e12614.
- Allison BJ, Brain KL, Niu Y, Kane AD, Herrera EA, Thakor AS, Botting KJ, Cross CM, Itani N, Skeffington KL, Beck C, Giussani DA. Fetal in vivo continuous cardiovascular function during chronic hypoxia. J Physiol. 2016; 594(5): 1247–64. <u>https://doi.org/10.1113/JP271091</u> PMID: <u>26926316</u>
- 41. Osol G, Mandala M. Maternal uterine vascular remodeling during pregnancy. Physiology (Bethesda). 2009; 24: 58–71.
- 42. Ducsay CA, Goyal R, Pearce WJ, Wilson S, Hu XQ, Zhang L. Gestational Hypoxia and Developmental Plasticity. Physiol Rev. 2017; 98(3): 1241–1334.
- 43. Kametas NA, McAuliffe F, Krampl E, Chambers J, Nicolaides KH. Maternal cardiac function during pregnancy at high altitude. BJOG. 2004; 111(10): 1051–8. <u>https://doi.org/10.1111/j.1471-0528.2004.00246.x</u> PMID: <u>15383106</u>
- McAuliffe F, Kametas N, Krampl E, Ernsting J, Nicolaides K. Blood gases in pregnancy at sea level and at high altitude. BJOG. 2001; 108(9): 980–5. PMID: <u>11563470</u>
- Ashwal S, Majcher JS, Vain N & Longo LD. Patterns of fetal lamb regional cerebral blood flow during and after prolonged hypoxia. Ped Research. 1980; 14: 1104–1110.
- Hecher K, Snijders R, Campbell S, Nicolaides K. Fetal venous, intracardiac, and arterial blood flow measurements in intrauterine growth retardation: relationship with fetal blood gases. Am J Obstet Gynecol. 1995; 173(1): 10–5. PMID: <u>7631665</u>
- Beckman JS & Koppenol WH. Nitric oxide, superoxide, and peroxynitrite: the good, the bad, and ugly. Am J Physiol. 1996; 271: C1424–1437. <u>https://doi.org/10.1152/ajpcell.1996.271.5.C1424</u> PMID: 8944624
- Clarke R, Daly L, Robinson K, Naughten E, Cahalane S, Fowler B, Graham I. Hyperhomocysteinemia: an independent risk factor for vascular disease. N Engl J Med. 1991; 324(17): 1149–55. <u>https://doi.org/ 10.1056/NEJM199104253241701</u> PMID: 2011158

- Lang CA, Mills BJ, Mastropaolo W & Liu MC. Blood glutathione decreases in chronic diseases. J Lab Clin Med. 2000; 135: 402–405. <u>https://doi.org/10.1067/mlc.2000.105977</u> PMID: <u>10811055</u>
- Kearney PM, Whelton M, Reynolds K, Muntner P, Whelton PK, He J. Global burden of hypertension: analysis of worldwide data. Lancet. 2005; 365(9455): 217–23. <u>https://doi.org/10.1016/S0140-6736(05)</u> <u>17741-1</u> PMID: <u>15652604</u>
- Psaty BM, Smith NL, Siscovick DS, et al. Health outcomes associated with anti-hypertensive therapies used as first-line agents: asystematic review and meta-analysis. JAMA 1997; 277: 739–45. PMID: <u>9042847</u>
- 52. Bourque SL, Gragasin FS, Quon AL, Mansour Y, Morton JS, Davidge ST. Prenatal hypoxia causes long-term alterations in vascular endothelin-1 function in aged male, but not female, offspring. Hypertension. 2013; 62(4): 753–8. <u>https://doi.org/10.1161/HYPERTENSIONAHA.113.01516</u> PMID: 23940196
- Peyronnet J, Dalmaz Y, Ehrström M, Mamet J, Roux JC, Pequignot JM, Thorén HP, Lagercrantz H. Long-lasting adverse effects of prenatal hypoxia on developing autonomic nervous system and cardiovascular parameters in rats. Pflugers Arch. 2002; 443(5–6): 858–65. <u>https://doi.org/10.1007/s00424-001-0766-9</u> PMID: <u>11889586</u>
- 54. Xiao D, Huang X, Xue Q, Zhang L. Antenatal hypoxia induces programming of reduced arterial blood pressure response in female rat offspring: role of ovarian function. PLoS ONE. 2014; 9(6): e98743. <u>https://doi.org/10.1371/journal.pone.0098743</u> PMID: <u>24905716</u>
- 55. Rook W, Johnson CD, Coney AM, Marshall JM. Prenatal hypoxia leads to increased muscle sympathetic nerve activity, sympathetic hyperinnervation, premature blunting of neuropeptide Y signaling, and hypertension in adult life. Hypertension. 2014; 64:1321–1327. <u>https://doi.org/10.1161/</u> HYPERTENSIONAHA.114.04374 PMID: 25267800
- Kitanaka T, Alonso JG, Gilbert RD, Siu BL, Clemons GK, Longo LD. Fetal responses to long-term hypoxemia in sheep. Am J Physiol. 1989; 256 (6 Pt 2): R1348–54.
- Herrera EA, Rojas RT, Krause BJ, Ebensperger G, Reyes RV, Giussani DA, Parer JT, Llanos AJ. Cardiovascular function in term fetal sheep conceived, gestated and studied in the hypobaric hypoxia of the Andean altiplano. J Physiol. 2016; 594(5): 1231–45. <u>https://doi.org/10.1113/JP271110</u> PMID: 26339865
- Benediktsson R, Lindsay RS, Noble J, Seckl JR, Edwards CR. Glucocorticoid exposure in utero: new model for adult hypertension. Lancet. 1993; 341(8841): 339–41. PMID: 8094115
- Celsi G, Kistner A, Aizman R, Eklöf AC, Ceccatelli S, de Santiago A, Jacobson SH. Prenatal dexamethasone causes oligonephronia, sodium retention, and higher blood pressure in the offspring. Pediatr Res. 1998; 44(3): 317–22. https://doi.org/10.1203/00006450-199809000-00009 PMID: 9727707
- Ortiz LA, Quan A, Zarzar F, Weinberg A, Baum M. Prenatal dexamethasone programs hypertension and renal injury in the rat. Hypertension 2003; 41(2): 328–34. PMID: <u>12574103</u>
- Dodic M, May CN, Wintour EM, Coghlan JP. An early prenatal exposure to excess glucocorticoid leads to hypertensive offspring in sheep. Clin Sci (Lond). 1998; 94(2): 149–55.
- 62. Figueroa JP, Rose JC, Massmann GA, Zhang J, Acuña G. Alterations in fetal kidney development and elevations in arterial blood pressure in young adult sheep after clinical doses of antenatal glucocorticoids. Pediatr Res. 2005; 58(3): 510–5. <u>https://doi.org/10.1203/01.PDR.0000179410.57947.88</u> PMID: <u>16148065</u>
- Newby EA, Myers DA, Ducsay CA. Fetal endocrine and metabolic adaptations to hypoxia: the role of the hypothalamic-pituitary-adrenal axis. Am J Physiol Endocrinol Metab. 2015; 309(5); E429–39. https://doi.org/10.1152/ajpendo.00126.2015 PMID: 26173460
- NICE Guidelines. Hypertension: Clinical management of hypertension in adults. <u>http://www.nice.org.uk/</u> guidance/cg127. 2011. Citation access date March 12 2018.
- Vasan RS, Larson MG, Leip EP, Evans JC, O'Donnell CJ, Kannel WB & Levy D. Impact of high-normal blood pressure on the risk of cardiovascular disease. N Engl J Med. 2001; 345: 1291–1297. <u>https://doi.org/10.1056/NEJMoa003417</u> PMID: <u>11794147</u>
- 66. SPRINT Research Group, Wright JT Jr, Williamson JD, Whelton PK, Snyder JK, Sink KM, Rocco MV, Reboussin DM, Rahman M, Oparil S, Lewis CE, Kimmel PL, Johnson KC, Goff DC Jr, Fine LJ, Cutler JA, Cushman WC, Cheung AK, Ambrosius WT. A Randomized Trial of Intensive versus Standard Blood-Pressure Control. N Engl J Med. 2015; 373(22): 2103–16. <u>https://doi.org/10.1056/ NEJMoa1511939</u> PMID: <u>26551272</u>
- 67. Bilardo CM, Hecher K, Visser GHA, Papageorghiou AT, Marlow N, Thilaganathan B, Van Wassenaer-Leemhuis A, Todros T, Marsal K, Frusca T, Arabin B, Brezinka C, Derks JB, Diemert A, Duvekot JJ, Ferrazzi E, Ganzevoort W, Martinelli P, Ostermayer E, Schlembach D, Valensise H, Thornton J, Wolf H, Lees C; TRUFFLE Group. Severe fetal growth restriction at 26–32 weeks: key messages from the TRUFFLE study. Ultrasound Obstet Gynecol. 2017; 50(3): 285–290. <u>https://doi.org/10.1002/uog.</u> 18815 PMID: 28938063

- Spencer R, Ambler G, Brodszki J, Diemert A, Figueras F, Gratacós E, Hansson SR, Hecher K, Huertas-Ceballos A, Marlow N, Marsál K, Morsing E, Peebles D, Rossi C, Sebire NJ, Timms JF, David AL; EVERREST Consortium. EVERREST prospective study: a 6-year prospective study to define the clinical and biological characteristics of pregnancies affected by severe early onset fetal growth restriction. BMC Pregnancy Childbirth. 2017; 17(1): 43. <u>https://doi.org/10.1186/s12884-017-1226-7</u> PMID: 28114884
- Dearden L, Bouret SG, Ozanne SE. Sex and gender differences in developmental programming of metabolism. Mol Metab. 2018; 15: 8–19. <u>https://doi.org/10.1016/j.molmet.2018.04.007</u> PMID: 29773464
- Sundrani DP, Roy SS, Jadhav AT, Joshi SR. Sex-specific differences and developmental programming for diseases in later life. Reprod Fertil Dev.2017; 29(11): 2085–2099. <u>https://doi.org/10.1071/RD16265</u> PMID: <u>28380326</u>
- Dasinger JH, Alexander BT. Gender differences in developmental programming of cardiovascular diseases. Clin Sci (Lond). 2016; 130(5): 337–48.
- 72. Gude NA, Broughton KM, Firouzi F, Sussman MA. Cardiac ageing: extrinsic and intrinsic factors in cellular renewal and senescence. Nat Rev Cardiol. 2018; 15(9): 523–542. <u>https://doi.org/10.1038/s41569-018-0061-5</u> PMID: <u>30054574</u>
- Ungvari Z, Tarantini S, Kiss T, Wren JD, Giles CB, Griffin CT, Murfee WL, Pacher P, Csiszar A. Endothelial dysfunction and angiogenesis impairment in the ageing vasculature. Nat Rev Cardiol. 2018; 15 (9): 555–565. <u>https://doi.org/10.1038/s41569-018-0030-z</u> PMID: <u>29795441</u>
- 74. Allison BJ, Kaandorp JJ, Kane AD, Camm EJ, Lusby C, Cross CM, Nevin-Dolan R, Thakor AS, Derks JB, Tarry-Adkins JL, Ozanne SE, Giussani DA. Divergence of mechanistic pathways mediating cardio-vascular aging and developmental programming of cardiovascular disease. FASEB J. 2016; 30(5): 1968–75. https://doi.org/10.1096/fj.201500057 PMID: 26932929
- Mendelsohn ME, Karas RH. The protective effects of estrogen on the cardiovascular system. N Engl J Med. 1999; 340(23): 1801–11. <u>https://doi.org/10.1056/NEJM199906103402306</u> PMID: <u>10362825</u>
- Rumbold A, Duley L, Crowther CA & Haslam RR. Antioxidants for preventing pre-eclampsia. Cochrane database system rev. 2008; CD004227.
- 77. Richter HG, Hansell JA, Raut S, Giussani DA. Melatonin improves placental efficiency and birth weight and increases the placental expression of antioxidant enzymes in undernourished pregnancy. J Pineal Res.2009; 46(4): 357–64. https://doi.org/10.1111/j.1600-079X.2009.00671.x PMID: 19552758
- Tain YL, Huang LT, Hsu CN. Developmental Programming of Adult Disease: Reprogramming by Melatonin? Int J Mol Sci. 2017; 18(2): pii: E426.
- 79. Niu Y, Kane AD, Lusby C, Allison BJ, Chua YY, Kaandorp JJ, Nevin-Dolan R, Ashmore TJ, Blackmore HL, Derks JB, Ozanne SE, Giussani DA. Maternal Allopurinol Prevents Cardiac Dysfunction in Adult Male Offspring Programmed by Chronic Hypoxia during Pregnancy. Hypertension. 2018; 72(4): 971–978. https://doi.org/10.1161/HYPERTENSIONAHA.118.11363 PMID: 30354714
- Murphy MP. Targeting lipophilic cations to mitochondria. Biochim Biophys Acta. 2008; 1777: 1028– 1031. <u>https://doi.org/10.1016/j.bbabio.2008.03.029</u> PMID: <u>18439417</u>
- Smith RA, Murphy MP. Animal and human studies with the mitochondria-targeted antioxidant MitoQ. Ann N Y Acad Sci. 2010; 1201: 96–103. <u>https://doi.org/10.1111/j.1749-6632.2010.05627.x</u> PMID: 20649545
- Graham D, Huynh NN, Hamilton CA, Beattie E, Smith RA, Cochemé HM, Murphy MP, Dominiczak AF. Mitochondria-targeted antioxidant MitoQ10 improves endothelial function and attenuates cardiac hypertrophy. Hypertension. 2009; 54(2): 322–8. <u>https://doi.org/10.1161/HYPERTENSIONAHA.109.130351</u> PMID: <u>19581509</u>
- Rodriguez-Cuenca S, Cochemé HM, Logan A, Abakumova I, Prime TA, Rose C, Vidal-Puig A, Smith AC, Rubinsztein DC, Fearnley IM, Jones BA, Pope S, Heales SJ, Lam BY, Neogi SG, McFarlane I, James AM, Smith RA, Murphy MP. Consequences of long-term oral administration of the mitochondriatargeted antioxidant MitoQ to wild-type mice. Free Radic Biol Med. 2010; 48(1): 161–72. <u>https://doi.org/ 10.1016/j.freeradbiomed.2009.10.039</u> PMID: <u>19854266</u>
- Snow BJ, Rolfe FL, Lockhart MM, Frampton CM, O'Sullivan JD, Fung V, Smith RA, Murphy MP, Taylor KM; Protect Study Group. A double-blind, placebo-controlled study to assess the mitochondria-targeted antioxidant MitoQ as a disease-modifying therapy in Parkinson's disease. Mov Disord. 2010; 25(11): 1670–4. <u>https://doi.org/10.1002/mds.23148</u> PMID: 20568096
- Gane EJ, Weilert F, Orr DW, Keogh GF, Gibson M, Lockhart MM, Frampton CM, Taylor KM, Smith RA, Murphy MP. The mitochondria-targeted anti-oxidant mitoquinone decreases liver damage in a phase II study of hepatitis C patients. Liver Int. 2010; 30(7): 1019–26. <u>https://doi.org/10.1111/j.1478-3231.2010.</u> 02250.x PMID: 20492507

- 86. Phillips TJ, Scott H, Menassa DA, Bignell AL, Sood A, Morton JS, Akagi T, Azuma K, Rogers MF, Gilmore CE, Inman GJ, Grant S, Chung Y, Aljunaidy MM, Cooke CL, Steinkraus BR, Pocklington A, Logan A, Collett GP, Kemp H, Holmans PA, Murphy MP, Fulga TA, Coney AM, Akashi M, Davidge ST, Case CP. Treating the placenta to prevent adverse effects of gestational hypoxia on fetal brain development. Sci Rep. 2017; 7(1): 9079. https://doi.org/10.1038/s41598-017-06300-1 PMID: 28831049
- Burnett RW, Covington AK, Fogh-Andersen N, Külpmann WR, Maas AH, Müller-Plathe O, Van Kessel AL, Wimberley PD, Zijlstra WG, Siggaard-Andersen O, Weisberg HF. Approved IFCC Recommendation on Definitions of Quantities and Conventions Related to Blood Gases and pH. Eur J Clin Chem Biochem. 1995; 33: 399–404.
- Vuilleumier J & Keck E. Fluorometric assay of vitamin C in biological materials using a centrifugal analyser with fluroescence attachment. J Micronutrient Analysis. 1998; 25: 25–34.
- Vatnick I, Schoknecht PA, Darrigrand R, Bell AW. Growth and metabolism of the placenta after unilateral fetectomy in twin pregnant ewes. J Dev Physiol. 1991; 15(6): 351–6. PMID: <u>1753075</u>
- Dawes GS, Johnston BM & Walker DW. Relationship of arterial pressure and heart rate in fetal, newborn and adult sheep. J Physiol. 1980; 309: 405–417 (1980). PMID: 7252872
- Itskovitz J & Rudolph AM. Denervation of arterial chemoreceptors and baroreceptors in fetal lambs in utero. Am J Physiol. 1982; 242: H916–920. <u>https://doi.org/10.1152/ajpheart.1982.242.5.H916</u> PMID: 7081461
- Herrera EA, Verkerk MM, Derks JB, Giussani DA. Antioxidant treatment alters peripheral vascular dysfunction induced by postnatal glucocorticoid therapy in rats. PLoS ONE. 2010; 5(2): e9250. <u>https://doi.org/10.1371/journal.pone.0009250</u> PMID: 20174656